



Systematic Review Multidisciplinary Design Automation of Electric Motors—Systematic Literature Review and Methodological Framework

Niklas Umland ^{1,*}, Kora Winkler ¹, and David Inkermann ²

- ¹ Fraunhofer Institute for Manufacturing Technology and Advanced Materials IFAM, Wiener Straße 12, 28359 Bremen, Germany; kora.winkler@ifam.fraunhofer.de
- ² Institute for Mechanical Engineering, Technische Universität Clausthal, Robert-Koch-Straße 32, 38678 Clausthal-Zellerfeld, Germany; inkermann@imw.tu-clausthal.de
- * Correspondence: niklas.umland@ifam.fraunhofer.de

Abstract: Electric motor development is a challenging task, as higher efficiency requirements and various interdependencies between different engineering domains must be considered. Established design approaches often lack the ability to address these interdependencies because they focus on specific domains and properties. Automated, multidisciplinary design approaches hold untapped potential for optimizing motors in terms of diverse requirements and advancing the development of more efficient and reliable motors. This paper presents a systematic literature review of the current state of research in the multidisciplinary design automation of electric motors. The literature basis comprises 1005 publications that are identified by a systematic internet search. The review of the existing approaches is based on twelve criteria that characterize the design automation task in general, such as knowledge representation or reasoning methods used, as well as criteria specific to electric motor design, such as domains considered and their coupling. The analysis reveals what current approaches are lacking: Consequent analysis and integration of domains, applicability of suggested methods, incorporation of established multidisciplinary design optimization (MDO) architectures, alongside the consideration of passive components in the motor. Aside from the introduction of twelve criteria for systematic charaterization of multidisciplinary design automation of electric motors, this article expands the state of the art by proposing an initial framework to establish process chains tackling the identified gaps in the review.

Keywords: electric motor; electric machine; design automation; multidisciplinary design optimization (MDO); knowledge-based engineering

1. Introduction

Electric motors account for 53% of electricity consumption worldwide [1]. They are prevalent in a wide range of industrial and consumer products. Research focusing on electric motor development has gained momentum due to the ongoing electrification of various mobile applications, e.g., electric cars and more electric aircraft, as well as an increasing amount of mechatronic products. Subsequently, an increasing effort is invested in processes and methods in electric motor development, to meet ever-increasing requirements with regard to application-specific efficiency and power density [2]. Most applications require electric motors to be custom-tailored, resulting in the need to provide more flexible and efficient development processes [3].

Typical engineering domains involved in electric machine design are illustrated in Figure 1. In conventional approaches, design decisions are made in the electromagnetic domain first, often relying on the designers' assumptions and experiences [4]. Requirements and restrictions of further engineering domains like rotordynamics, structural mechanics or thermal design are considered at a later stage. Every time a design does not meet these



Citation: Umland, N.; Winkler, K.; Inkermann, D. Multidisciplinary Design Automation of Electric Motors—Systematic Literature Review and Methodological Framework. *Energies* **2023**, *16*, 7070. https://doi.org/10.3390/ en16207070

Academic Editors: Zbigniew Hanzelka, Grzegorz Sieklucki and Tomasz Drabek

Received: 15 September 2023 Revised: 5 October 2023 Accepted: 10 October 2023 Published: 12 October 2023



Copyright: © 2023 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). requirements, several steps or the whole design process have to be revisited [5]. This often results in lengthy iteration cycles, increased project risks and prevents the inclusion of aspects from different domains in decision making. Various authors claim this to be a key factor for future electric motor development to meet rising efficiency standards [6–9].



Figure 1. Engineering domains involved in the design of electric motors.

For this reason, new design methods are suggested in the literature, which will be summarized under the term multidisciplinary design automation (MDDA) in this work. These aim to overcome the weaknesses of conventional design approaches by systematically searching the solution space through automation. This allows decision-relevant information to be made available earlier and, thus, allowing for the design of a better electric motor from a multidomain perspective by breaking down domain silos [10,11].

In the literature, different approaches can be found for the application of MDDA in the development of electric motors. However, there is a lack of higher-level methods to set up MDDA processes. This paper systematically reviews the research conducted on MDDA in electric motor engineering. The main objective is to analyze and classify existing methods in this area. This allows to identify further research needs. Overall, this paper contributes by (1) presenting the state of the art regarding the adoption of MDDA, (2) identifying trends and gaps resulting from it and (3) deriving a generic methodological framework for the adoption of MDDA.

The article is structured as follows: Section 2 gives an overview of the used terminology in the scope of MDDA; Section 3 describes the review method; and Section 4 presents the employed review criteria. Section 5 presents the results of the systematic literature review. Section 6 presents the findings of the review in order to identify trends and gaps. Based on these findings, the authors present a methodological framework for setting up MDDA in electric motor engineering in Section 7. Section 8 concludes the paper by answering the research questions.

2. Multidisciplinary Design Automation in Electric Motor Design

2.1. Terminology

The most relevant terms for this publication will be clarified in the following text, since there is no uniform terminology. In general, design automation (DA) can be understood as computer-aided engineering support. It translates information and knowledge into solutions, tools or systems to support the progress of design processes through automation [12]. Since electric motor development involves different engineering domains, it is reasonable to extend the concept to multidisciplinary design automation as suggested by [13]. Another term that is often used in the context of DA is knowledge-based engineering (KBE). Van der Velden differs between KBE and DA and describes the former as a capture of engineering knowledge to emulate human decision making and automate engineering processes. Design automation is described as an automation of sequential steps in engineering processes. These are often only applicable to specific use cases, since they contain hard-coded rules and knowledge [14]. In more recent definitions, KBE has been described as a method within the context of DA [13]. These definitions emphasize that KBE leverages existing knowledge to automate repetitive tasks in product development [15,16]. As shown in Figure 2, three steps are necessary for the creation of a KBE system: knowledge capture, knowledge formalization and knowledge representation [17].



Figure 2. Steps necessary to build a KBE system [17].

Generative design and engineering is a term that is often used synonymously to DA. It describes methods and processes for generating a large number of solution variants, often related to geometries. This enables engineers to explore the existing solution space more thoroughly. There are different approaches to generative design, ranging from parameterized computer-aided design (CAD) templates [18] to leveraging deep networks to synthesize new designs [19]. The term computer-based design synthesis (CDS) is used in a similar context. It refers to tools and methods in the design process that support engineers to create design alternatives based on computationally encoded knowledge [10,20]. Another relevant term in the context of automated multidisciplinary design is multidisciplinary design optimization (MDO). The concept of MDO emerged primarily from the challenges encountered in aircraft design, which inherently involves intricate design and simulation processes due to its multidisciplinary nature. MDO addresses these challenges by executing coupled simulations and concurrent optimization of specified design variables [21]. In recent years, MDO methodologies have been applied to the design of electric motors [22,23]. Since MDO facilitates the automatic synthesis of geometries while considering multiple domains of analyses, it can be regarded as a distinct form of MDDA that leverages optimization algorithms as a reasoning approach. A typical process for setting up a MDO consists of five steps, as visualized in Figure 3 [24].



Figure 3. Steps for the creation and execution of MDO processes [24].

Design automation tasks serve different purposes in the design process. In his categorization—see Figure 4—Rigger groups DA task categories along the design process. Some of them contain multiple tasks, resulting in a total of 15 different design automation task categories. In early design stages, design automation tasks are focused on automatically decomposing the functional structure, i.e., functional synthesis, and on product architecture synthesis. This includes topology and parameter synthesis and analysis. However, at this early stage, the design takes place on a high level and does not yet include concrete geometric features and parameters. In embodiment design, tasks revolve around spatial synthesis and analysis and are grouped under the term spatial

product/component architecture. These range from topology and parameter synthesis to analysis preparation, integration and simulation. A spatial topology optimization forms an integrated loop of these tasks. Typical MDO processes are an example of this. Automated design configuration can be applied both in embodiment design and detailed design. Tasks grouped under the term transformation process design focus on the planning of production processes in later stages of the design process. However, a co-evolution of products and processes can be implemented in early stages too [10].



Figure 4. Categorization of design automation tasks along the design process [10].

Apart from the selection of a motor type, most decisions made in electric motor design are made during spatial product design. The focus of this review is therefore on design tasks involving an automated synthesis of geometries, i.e., spatial product/component architecture, design configuration and spatial topology optimization.

2.2. Existing Reviews

In order to ensure that there is no review already addressing the research goal, existing review papers in the field of MDDA for electric motors were searched in the lens.org database. The identified review papers are briefly presented in the following text.

Reference [2] gives an overview of modern optimization techniques within the development of electric machines. The authors present commonly used optimization algorithms and the basic idea of setting up the optimization problem. However, integrated multidisciplinary analyses are not in focus.

Reference [25] differentiates between component-level design and system-level design approaches and argue that the optimal solution can only be found following a multidomain perspective. Furthermore, aspects to be considered in each of the engineering domains, i.e., electromagnetic, thermal and mechanical engineering, are listed. However, no in-depth analysis of the state of the art in MDO for electric machines is presented.

Reference [26] highlights and categorizes couplings between engineering domains in the multidisciplinary optimization of electric machines. The proposed categorization was applied within the framework of the systematic review in this paper; see Section 4.3.

References [27,28] review current optimization procedures used within electric machine development, focusing on optimization algorithms. Since a profound analysis of optimization algorithms has already been carried out, the analysis of these algorithms will only be a secondary aspect in the paper at hand. In the systematic analysis, the associated data collection category is based on the categorization proposed by [27]. To summarize, none of the existing review papers comprehensively address the specific use of MDDA for electric motor development. Single reviews focus only on certain aspects, e.g., optimization algorithms, while in other reviews, MDDA is only considered as a side aspect.

2.3. Research Objectives and Focus

MDDA is increasingly relevant to address the need for more efficient development of electric motors, especially as solutions are required to meet increasing demands for power density, reliability or efficiency. Based on the experience gained in different projects at Fraunhofer IFAM, there is a need for a methodology for setting up application-dependent MDDA workflows. In order to provide a sound basis for this methodology, the research goal is formulated as follows:

Purpose	Collect and analyze existing approaches
Issue	on MDDA in electric motor development that explicitly take into account
Object	different engineering domains and their interactions
Perspective	both from the point of view of engineers and researchers.

Based on this goal, this article is guided by the following overarching research question:

What approaches for MDDA exist to support development of electric motors, which domains do these consider, what activities and methods do these include and in which application fields are they employed?

Since existing reviews do not answer this question, the contribution at hand extends the state of the art by

- Collection and analysis of state of the art MDDA in electric motor development.
- Introduction of criteria to distinguish MDDA workflows and associated methods.
- An initial framework to set up MDDA workflows for efficient development of electric motors for different applications.

These contributions are based on a systematic literature review. In line with the design research methodology (DRM) [29], this review is associated to the "descriptive study 1". It serves as a basis for detailed design support development in the prescriptive study in future work.

3. Method

The method of the systematic review is introduced in the following section. The applied methodology is based on [30], including the following steps: (1) research questions, (2) search process, (3) publication selection, (4) quality assessment, (5) data collection and analysis. For publication selection the PRISMA flow was adopted [31].

3.1. Research Questions

Based on the understanding of MDDA described in Section 2.1 and the DA tasks that are in focus in this paper, the following sub-questions are used to guide the review:

- 1. How is the multidomain analysis and automated geometry synthesis structured?
- 2. How are domain-specific aspects and constraints formalized?
- 3. When in the development process of electric motors are MDDA approaches being utilized?
- 4. For which application and type of electric motor is MDDA already employed?
- 5. Which parts and geometries of electric motors are automatically created?
- 6. Which domains are considered and implemented in the MDDA and at which fidelity level?
- 7. Which couplings between engineering domains are considered?

Based on these sub-questions, specific criteria for the review will be defined in Section 4.

3.2. Search Process

To ensure completeness of papers for a systematic review, the authors searched for relevant publications in a two-stage process, as depicted in the PRISMA diagram in Figure 5. In the first stage, the search was based on a search string, using the search databases lens.org, Web of Science and Scopus. These offer a large database and are able to handle complex search strings. In an iterative process, the search string was specified, considering three parts and their multiple synonyms: design automation, multidisciplinary, and electric motor. The finally used search string is given in Appendix A. The search was conducted within title, abstract and keywords of the publications. In the next steps, doublets were removed, inclusion/exclusion criteria were applied, and the database was extended by a forward and backward citation analysis. In Figure 5, the different steps of the literature identification and selection process following the PRISMA flow diagram [31] are illustrated.



Figure 5. PRISMA flow of the conducted literature identification process and number of papers in each step.

3.3. Publication Selection

Following criteria were used to select papers for the review:

- Document type: Peer-reviewed conference or journal publications
- Language: English
- Date range: Years 2007–2022
- Electric machines: Only papers that focus on design activities of electric machines are included. Publications that consider electric machine design as a side aspect (system-of-systems) are not considered.
- Electric motor focus: Papers that relate to details of generators are excluded from the review unless the authors indicate the transfer to electric motors.
- Automated geometry synthesis: Only papers that employ methods for automated geometry synthesis are considered.
- Multidisciplinary: In the scope of automated geometry synthesis, analyses have to be presented in more than one engineering domain, see Figure 1.

A total of 934 papers were initially identified, as depicted in Figure 5. After removing duplicates, 634 papers remained for abstract screening. In this screening process, 437 papers were excluded. Four papers could not be retrieved and therefore had to be excluded from

the review. Another 122 papers had to be excluded in the assessment of eligibility based on the full papers, either because they did not fit the inclusion criteria or turned out to be duplicates. In this step, publications are classified as duplicates if they only differ marginally in their content. This is often the case when publications were first published at conferences and then in a journal. In this case, the more detailed version of the paper is considered in the review. A forward and backward citation process was conducted based on the remaining studies. In this process, 71 additional papers were identified based on their abstract. All 71 papers could be retrieved and thus were assessed for eligibility, which resulted in an exclusion of 48 papers. As a result, a total of 94 papers were included in the review.

3.4. Quality Assessment

Three basic aspects were considered to assure high quality of the review during the literature selection phase:

- 1. Quality of paper identification;
- 2. Quality of papers;
- 3. Quality of review criteria

Moreover, the overall procedure of the review is based on well-known methods, i.e., Kitchenham's approach and PRISMA flow [30,31]. In order to address the three mentioned quality aspects, the following measures were taken. Literature identification was based on three well-established databases, namely Lens.org, Scopus and WebOfScience. A backward and forward citation analysis based on Researchrabbit and lens.org was conducted to assure that all relevant papers were found for the review. To ensure the quality of papers in the study, only conference and journal papers having passed a peer-review process were included. The quality of the review criteria—see Section 4—is ensured by two precautions. First, categories and items are based on existing works. Second, within two randomly selected samples (29 papers), the classification of papers was performed by two independent authors. Based on the results, discrepancies were identified and the review criteria were specified to reduce subjectivity of classification. The occurring discrepancies were divided into three categories: full match of review items, partial match and poor match. As a result of the iterative procedure, the average for all review criteria was 75.5% full matches, 15.3% partial matches, and 9.2% poor matches.

3.5. Data Collection and Analysis

To systematically analyze the papers and answer the research questions, twelve criteria were defined for the review; see Figure 6. Half of these criteria are based on [10] and extended by specific criteria relevant for design automation in electric motor development. For the definition of basic DA tasks, Rigger analyzed publications regarding inputs, outputs, goals and DA Methods. The analysis of goals is further split up in purpose, system level and requirements/constraints, while DA Methods consist of knowledge representation and reasoning methods. Since the review at hand focuses on MDDA for electric motors the following criteria were added for the review: considered domains, couplings between domains, machine type, part of the machine, and application. Considered domains were analyzed based on [32], couplings were introduced based on the classification in [26], and reasoning methods were adapted for electric motor specific reasoning methods defined in [27].



Figure 6. Review criteria employed in the review.

4. Review Criteria

The chapter below describes the criteria used for the systematic review and introduces the information to be extracted for each criterion.

4.1. Inputs, Outputs and Goals

The inputs and output types of a design automation task depend on the stage within the design process [10]. In this review, we distinguish between parameterized and nonparameterized geometry, as proposed by [2]. Instead of employing a fixed topology defined by parameters, the corresponding approaches utilize a non-parameterized geometry that discretizes design areas and achieves design optimization by element-wise material assignment. Furthermore, alternative geometry inputs, such as component libraries, were explored during the search process. An expected output of spatial synthesis tasks is finding (optimized) geometry parameters [10]. The goal criterion provides additional information regarding the purpose of geometry synthesis and its objective function. In this review, the authors assessed whether topology or parameter synthesis was performed, as well as the objective function and design variables employed in each respective paper. Given the multitude of potential design variables, specific variables were not identified. Instead, the review identified the overarching component they represented. For example, when stator slot parameters were included as design variables, the review noted the term "stator".

To define the process stage in which the MDDA is applied, the specific design variables were analyzed. Allocation of design variables to process stages of electric motor development is defined based on common guidelines, e.g., [32,33]. In both procedures, the first step is to determine the main machine dimensions, namely the air gap diameter and the active stack length. In order to determine the stage of the development process, this review evaluates whether the main machine dimensions were already fixed or considered as design variables in the MDDA. The former was defined as detailed design and the latter as preliminary design. Some papers describe a two-stage design process, where a preliminary design is found in the first stage and further elaborated upon. These papers were marked as preliminary and detailed design in the study.

4.2. Knowledge Representation and Reasoning Methods

Rigger highlights that the selection of knowledge representation methods in the context of design automation is influenced by the particular reasoning approach used. Hence, the review analyzed the way of representing knowledge based on the characteristics proposed by [10]. These include: graphs, object-orientated representations, ontologies, shape-based representations and procedural rules. The identification of reasoning methods was guided by Rigger's definition, while specific reasoning methods relevant for electric machines, as defined in [27], were added. In total, seven categories of reasoning methods were defined:

- 1. Evolutionary algorithms, e.g., genetic algorithms (GA);
- Other intelligent optimization algorithms, e.g., ant colony optimization algorithm (ACO);
- 3. Multi-objective optimization algorithms, e.g., non-dominated sorting genetic algorithm (NSGA);
- 4. Gradient-based algorithms, e.g., sequential quadratic programming (SQP);
- 5. Inference engines, i.e., fixed decision procedures to guide rule application;
- 6. Statistical reasoning methods, e.g., response surface method (RSM);
- 7. Other reasoning methods.

4.3. Considered Domains and Domain Couplings

In this study, publications were only included when their automated geometry synthesis process follows an in-the-loop MDA; see Figure 7.



Figure 7. Difference between in-the-loop multi-disciplinary analysis (MDA) (**a**), where multidisciplinary aspects are considered in decision making and multi-disciplinary (MD) feasibility check (**b**), where a primary discipline is used for decision making, and other domain analyses are conducted to ensure the design's feasibility afterwards.

In conventional electric motor development processes, the design is optimized in one domain, mainly the electromagnetic domain. This single-domain optimum is then checked in other domains to ensure its feasibility. The term in-the-loop MDA, however, pertains to process frameworks that assess designs across multiple engineering domains for each design variant. This approach empowers engineers to make design choices by considering properties from multiple domains and facilitates more frequent iterations, thereby facilitating the identification of optimal designs within the solution space [6,7]. The review papers were analyzed regarding the following engineering domains considered in the in-the-loop MDA: electromagnetic, thermal, mechanic, manufacturing and economic domains. A variety of different engineering tasks can be summarized under the term mechanical design [32]. This is why it was decided to further split this domain in the study into rotordynamics, structural and NVH (noise, vibration, harshness) analysis.

In addition to determining the relevant domains to consider, setting up MDDA involves making decisions regarding the appropriate model fidelity. In this review, three fidelity levels across the different domains were specified. These are analytical calculations, lumped-parameter models (LPM), and finite element analyses (FEA) or computational fluid dynamics. Models with lumped parameters are very popular in the design of electrical machines to evaluate the thermal behavior of the machine. However, they can also be used to evaluate other physical relationships, such as noise characteristics [34].

These domain models are inherently coupled because they are based on the same geometry and material properties. In addition to that, the results of one domain model can affect the other model. To handle these influences, couplings between the different domains were analyzed in the review. These couplings in electric machine design processes describe the sharing of input and output data between different analyses [26].

There are different ways to handle domain couplings. The first one is to separate the analysis modules and not consider a coupling. The second one is to sequentially couple domains, i.e., use results of one domain in the other domain in a one-way fashion, called one-way coupling in this review. The third one is to consider a domain coupling in an iterative manner, meaning that the inputs and outputs of both domains are iterated until the error between inputs and outputs reaches a certain threshold. The latter will be called two-way coupling in this review. A special type of coupling is the so-called parameter split decoupled approach. In this approach, parameters are assorted to domains based on their sensitivity in order to reduce the impact of decoupling [26].

4.4. Type and Part of Electric Motor and Application

To assess the scope of the MDDA, the automatically designed geometries of the electric motor are noted in the review. Furthermore, the type of electric motor, e.g., permanent magnet synchronous motor (PMSM), is recorded. In order to find areas of focus for the use of MDDA, application fields were identified when indicated by the authors.

4.5. Validation

To evaluate the credibility of the proposed methods, the type of validation was also identified. This involved determining whether validation was conducted through simulations or experiments within the paper, after discovering a specific design using MDDA.

The described review criteria were applied to all publications and the classification was performed by one author.

5. Results

In this section, we summarize the results of the literature review.

5.1. General Results

With regard to the evolution of MDDA as well as the validation of MDDA approaches, these are the following results.

5.1.1. Trend

In Figure 8a, the amount of papers for each year is given. The graph shows that over the analyzed time span of 15 years (2007–2022), the amount of publications regarding MDDA in electric motor development is constantly increasing. About 56% of the papers considered within the review were published in the last 5 years (2018–2022). Most of these papers are published by researchers, while only a small portion (30%) of the works include at least one author affiliated to an industrial company.





(a)

(b)

Figure 8. (a) Number of papers in the review by year published; (b) Share of papers with different types of validation in the review.

5.1.2. Validation

Figure 8b reveals that the majority of papers included at least one kind of validation. The highest form of validation is to use high-fidelity simulations and prototype tests to verify the MDDA results. The type of validation depends on the analyses conducted, i.e., if the MDDA included high-fidelity simulations of one specific domain, validation can only be performed by even higher fidelity simulations or experimental results. In [8], an FEA-based validation is conducted for each analysis module. Reference [35] compares the results of their optimization workflow with an analysis in the software Ansys Motor-CAD for validation. In [36], the applied design process is verified by applying it to an example and comparing it to detailed FEA analyses.

In [37], the design found by the MDDA is prototyped and tested physically. The results are close to simulation results, whereby the authors consider the proposed development methodology to be validated. Reference [11] validates their proposed MDDA tool in a similar manner by comparing properties of the realized prototype with analysis results.

5.2. Inputs, Outputs and Goals

5.2.1. Inputs

In this review, most papers used parameterized geometries as inputs for the MDDA. However, the level of geometry fidelity differs. Approaches that apply analytical analyses utilize a geometrical representation of their design variables, whereas the actual calculations are performed using only the parameters (e.g., [36,38,39]). Higher fidelity geometry representations, such as CAD models, are primarily employed for more detailed analyses, such as FEA simulations. In these papers, the CAD representation is generated and used for the analysis (e.g., [40,41]). While most authors use a parametrized geometry, some authors adapt non-parametrized geometries, enabling them to find new and unconventional designs. For instance, Reference [42] uses a level-set methodology to describe the geometry of a PMSM rotor and analyzes it in the thermal and electromagnetic domain. A similar approach is proposed by [43] for the joint electromagnetic and mechanical design of a rotor geometry for a synchronous reluctance motor.

5.2.2. Goals

Consistent with the inputs, the goal of most of the contributions is a parameter synthesis. It is noticeable that almost no paper analyzes different topologies within the MDDA. Exceptions are the papers applying non-parametrized geometries, since these will always result in new topologies. Furthermore, Reference [44] compares various rotor

topology concepts for Synchronous Reluctance Motors (SynRM) that differ based on the positioning of supporting ribs and the use of a retaining sleeve. Similarly, Reference [11] uses a MDDA process to compare different PMSM topologies. In both cases, however, the optimization is performed separately for each topology. The results of each MDDA is then compared, leaving the topology decision to the machine designer.

5.2.3. Outputs

Since all MDDA approaches focus on geometry synthesis, geometry was identified as an output for all papers in the review. However, the aspects under which the design decisions or optimizations were conducted differ. These objectives describe a design's properties that the MDDA aims to maximize or minimize. The most common synthesis objectives are shown in Figure 9a.



(a)

Figure 9. (a) The ten most frequently identified electric motor properties as a part of the objective function in each paper; (b) Share of papers with different types of reasoning methods in the review.

5.2.4. Objectives

A total of 21 papers employ a single objective, implying that most papers use a multiobjective function in their MDDA. From Figure 9a, it is apparent that the three most prominent objectives are the main performance objectives of electric machines, namely torque, efficiency, and power. These originate directly or indirectly from the electromagnetic analysis of the proposed design. Cost and mass can mainly be derived from a machines' geometry [45,46]. However, their calculation depends on the considered geometries and their fidelity. The majority of papers only take into account active parts for cost and mass calculation. Approaches that consider non-active parts for cost or mass calculations are only available in [11,46]. However, these examples demonstrate that the mass estimate for non-active parts is by no means negligible, as they account for 40% of the total mass of the machine. Losses in an electric motor consist of iron losses, winding losses and mechanical friction losses [47]. In some works, losses act as an indirect way of taking thermal aspects into account [48]. Some papers even abstain from adding a thermal analysis module for this reason. However, doing so prevents assessing temperatures and results in not fully exploiting a material to its limits [39]. Noise, temperature, stress and torque ripple are, again, domain-specific properties of a machine, stemming from NVH, thermal, structural and electromagnetic analyses.

5.3. Knowledge Representation and Reasoning Methods

5.3.1. Reasoning Methods

Figure 9b shows the reasoning methods employed in this review. Most reasoning methods in this review rely on optimization algorithms. Rule-based reasoning using inference engines can only be found in a small number of papers, e.g., [38,49]. Evolutionary algorithms make up a large proportion of reasoning methods in the scope of this review,

with the genetic algorithm (GA) being most widely used [50–52]. Other intelligent algorithms, like the particle swarm optimization (PSO) method, are used less frequently, e.g., [53,54]. Most optimization algorithms have further developed counterparts to handle multi-objective optimizations better, e.g., the widely used non-sorting genetic algorithm (NSGA-II) or multi-objective PSO (MOPSO) [40,55,56].

5.3.2. Knowledge Representation

According to [10], the formalization of knowledge in the context of a general design automation task is strongly related to the reasoning methods used. In the reviewed papers, design dependencies were almost exclusively formalized as being shape-based or via procedural rules. As introduced in Section 2, the design automation tasks of the reviewed papers fall under the category of spatial product synthesis. For these tasks, typical knowledge representations next to shape-based and procedural formalization are graph-based formalization, object-oriented formalization and formalization based on the unified modeling language (UML) [10]. However, in the papers reviewed, these formalizations were not identified.

5.4. Considered Domains and Domain Couplings

Figure 10b shows how many different disciplines are considered in the MDDA of the reviewed papers. As it can be seen, most approaches are limited to two domains. Figure 10a illustrates the frequency of each domain and the corresponding fidelity level identified in this review.



(a)

Figure 10. (a) Considered domains in the papers of this review and corresponding fidelity level; (b) Share of papers considering two, three or four and more domains. For this electromagnetic, thermal, structure mechanic, rotordynamic, NVH, manufacturing and economic analyses as part of the MDDA were searched for.

5.4.1. Electromagnetic Domain

From Figure 10a, it is apparent that nearly all reviewed papers consider the electromagnetic domain. This emphasis is primarily driven by the opportunity to calculate machine efficiency via electromagnetic models based on the calculation of electromagnetic losses and the machine power from torque and speed [57]. These parameters are crucial while designing electrical machines and explain the number of studies in Figure 9a that contain torque, efficiency and power as optimization objectives [58]. Some papers additionally mention the analysis of torque ripple and cogging torque [47], whose reduction together with keeping a minimum efficiency can also be a desired optimization criteria in the MDDA of electrical machines [59]. Overall, in this review, three major distinguishing criteria for the electromagnetic analysis were observed: analysis fidelity level, analysis dimension (2D/3D) and time dependency of the analysis. Within the electromagnetic domain, highfidelity approaches prevail. In that case, most of the authors use numerical FEA concepts, like those demonstrated in [60,61]. Other numerical approaches for the electromagnetic analysis, e.g., using the finite-difference method, were not identified in this review. Next to numerical approaches, analytical models [48,54,62] and lumped-parameter models (often referred to as equivalent circuit approaches) [34,57] are also used in the reviewed papers. The required level of fidelity for the electromagnetic analysis depends on the machine type under investigation and possible compromises between computational effort and accuracy needed for the desired outcomes [63]. For instance, FEA has a high fidelity but entails a high computational burden. In contrast, analytical or equivalent circuit approaches have moderate computational effort but offer lower fidelity [64]. Nevertheless, in case of difficult structures, complex field distributions and nonlinearities, analytical models may fail to predict electromagnetic variables or are not accurate enough, necessitating the use of FEA despite the computational demands [47,65]. A possible way to reduce computational effort while keeping high fidelity in the electromagnetic analysis is the combination of different methods. In [47], an approach is proposed where parameters that are easier to model are analyzed using an analytical sub-domain model; then, the more challenging parameters are optimized using an FEA. The analytical model optimizes the PMSM's general geometry by calculating its electromagnetic and torque performances with high accuracy. To further optimize the torque ripple and cogging torque, the authors utilize FEA in computing the magnetic pole shift, magnet segmentation and eccentricity. On the other hand, the combination of an analytical approach and FEA can be applied when the analytical part is used for post-processing of loss data obtained from FEA [66]. In [57], they present a combination of a magnetic equivalent circuit (MEC) for the electromagnetic domain and an electric equivalent circuit (EEC) for the electrical domain and as extension of a loss model for calculating the copper and iron losses. In combination, these models are used to replace an FEA for the electromagnetic optimization of a switched reluctance machine. To calculate the static flux density, the static flux linkage characteristic data and the static torque, a magnetic circuit modeled by a reluctance network is applied. Figure 11 demonstrates an example for a reluctance network. The EEC is then used to calculate, for instance, the phase current, the torque ripple coefficient and the average torque from the static torque and the flux linkage characteristic data [57]. A similar way is chosen in [67], where a strategy is proposed to mitigate computational effort by substituting the FEA with an analytical method to calculate the torque, internal power factor and the internal voltage and combining that method with an EEC and a loss model. The choice to use a combination or a single EEC and MEC depends on the specific use case and the desired parameters as well as on whether only the electromagnetic or the entire electrical analysis is considered, which is also related to time dependency [63]. If a static model is sufficient to predict the behavior of a machine, as is the case for synchronous machines, the electromagnetic and electrical analyses can be combined in an FEA or MEC without requiring an extra EEC [63]. In this context, static or quasi-static analysis means an analysis for one or more static rotor position points [68], which results in lower computational effort [45] and faster analysis if it is used to replace a transient simulation [38]. However, in general, transient analysis serves to more accuracy [69] and is also utilized to analyze transient behavior caused by eddy currents [70]. Furthermore, the need for a transient analysis depends on the machine type. Induction machines, for instance, require transient analysis or the analysis needs to run until achieving a steady state [63]. This requirement leads to a significant increase in computational effort. Hence, for dynamic or transient performance analysis, an electrical analysis model such as an EEC is imperative to limit the computational effort [63]. The combination of an analytical model and a MEC is described in [36]. The authors utilize the methods separately, employing one for the stator field calculation aspects and the other for electromagnetic parameters relating to the rotor. Another approach to reduce the computational effort involves a combination of methods, such as implementing an

analytical coarse pre-optimization followed by a more detailed FEA. In [71], this technique is used to analyze the electromagnetic characteristics of a PMSM. A similar approach is employed in [72]. The method described in [73] is comparable but incorporates a MEC for achieving a low fidelity and low computational effort. In [64], all three fidelity levels are combined. The analytical model is used for a pre-optimized design to filter out weak options. This is followed by a MEC analysis. The final accurate optimization is then made by an FEA.



Figure 11. Example of a magnetic equivalent circuit (modified from [57]).

5.4.2. Thermal Analysis

The thermal domain is considered in about two thirds of reviewed papers; see Figure 10a. The purpose of the thermal calculator is to assess whether design candidates reach thermal limits [50,59]. Typical thermal limits are the maximum winding insulation and magnet deterioration temperature [22,45,59]. As it can be seen in Figure 10a, fidelity level 1 is applied less frequently for thermal calculations. The authors in [38] use analytical equations to derive cooling jacket dimensions and a heat exchange coefficient in the scope of a multidisciplinary automated scaling approach. The most prominent way of conducting thermal analyses are lumped-parameter models. Their execution times are usually very fast; however, it requires the developer to build a model to accurately describe the main heat flow [22].

For this purpose, the machine geometry is modeled by simple geometrical shapes for which the heat flow has been derived. Each of these are represented as a node in the LPM [56,74]. Heat flow between nodes are modeled using thermal resistances. For transient considerations, the heat capacity in each node must be taken into account [56]. Three different detailing criteria of LPMs were observed in this review: (1) The dimensionality, (2) time dependency and (3) assumptions for the thermal modeling. Most authors use a 2D representation of the machine for the thermal analysis by modeling an axial or radial section of the motor [56,75,76]. Three-dimesional LPMs combine these and have a higher accuracy [72,77]. Stationary thermal calculations are most often used in the papers of this review, i.e., calculation of temperatures under the assumption of reaching an equilibrium in each considered operating point [36,75]. However, not considering transient phenomena with respect to use scenarios can result in a significant limitation of the design space. Reference [39] compares in his MDDA application steady-state and transient thermal analyses for the design of an electric motor in a hybrid electric aircraft powertrain. The use of a transient thermal analysis with respect to the flight scenario results in a significant size reduction and thus an almost threefold power density increase. Some other detailing criteria of LPMs are the assumptions made in each study. Commonly, radiation heat transfer is not considered, and thermo-physical material properties are assumed to be temperature-independent [39,57,78]. However, more detailed LPMs encompassing radiation heat transfer in the scope of MDDAs

exist [72]. Beyond that, higher fidelity numerical modeling for thermal analyses is usually avoided for computational efficiency reasons.

It is noticeable that there are only a few papers that not only examine the exceeding of thermal limits in the context of an MDDA, but also include components and parameters of the cooling system as part of the design variables. One example is the design of cooling ducts in the scope of MDDAs [79,80]. In [81], the thermal stress distribution in a double-skewed rotor for an induction motor is used to find the optimal design of the rotor.

5.4.3. Structural Mechanics

The overarching objective of structural analyses in MDDAs is to ensure that the geometry is able to withstand the mechanical loads that are imposed during operation. For the most part, structural analyses are focused on rotor geometries, including rotor shaft, rotor sleeves, magnets and machine-specific rotor lamination geometries, like flux barriers in synchronous reluctance machines. The structural-mechanical design is often in conflict with the electromagnetic design, e.g., when sizing ribs in SynRM rotors [75,82]. Structural analyses in this review utilized both analytical calculations and numerical FE calculations, as illustrated in Figure 10a. In order to perform an analytical calculation of the structural mechanics within a MDDA, authors utilize model reductions for which analytical calculation models are available, i.e., reducing the problem to two dimensions by assuming the stress and strain in the axial direction to be zero [70]. In [37], an analytical model based on rotating disks stress equilibrium theory is applied for the design of a high-speed PMSM. The contact pressure, the thermal expansion as well as the centrifugal forces are considered. Similar approaches to design rotor magnets and rotor sleeves are adopted in [8,39,55,73,75]. Some authors use analytical calculations for more detailed rotor and stator geometries as well, e.g., ribs in SynRM rotors by employing beam theory simplifications [36]. Since this form of geometry simplification is limited when dealing with more complex geometries, FEA calculations tend to be used for this purpose in the reviewed papers. Thus, there are a variety of papers that use 2D FEA simulations to evaluate detailed rotor geometries in synchronous reluctance machines [44,68,83] and permanent magnet-excited machines [22,43,66,84]. For the design of high-speed rotors that include retention sleeves, 2D FEA is also utilized [50,60]. Regardless of the fidelity of the calculation, the maximum speeds including a safety factor are generally used as a basis for the structural mechanics analysis within an MDDA.

5.4.4. Noise, Vibration, Harshness (NVH)

Another set of tasks that can be assigned to the mechanical domain involves the examination of generation of acoustic noise. In this review, a total of 17 papers considered NVH analyses. Radial electromotive forces of different spatial orders and temporal frequencies cause the stator teeth to vibrate. Resonance amplifies this effect, which occurs when the spatial orders and temporal frequencies of the radial electromagnetic forces align with a natural mode of the considered geometry, resulting in a significant noise level [85,86]. The vibration in the stator is transmitted to components that are rigidly connected to the stator, such as the housing [41]. Usually, only the stator is considered in NVH analyses, although the housing and other motor components can certainly have a dampening effect [87]. In [85] the stator is simplified as a cylinder so that a mass-spring equation system can be set up for each mode shape. Similar analytical analyses are conducted in [22,88,89]. In [41], an equivalent circuit model is used to account for the different areas and materials of the stator and housing in the vibration analysis. The main advantage of these analytical and equivalent circuit approaches is the significantly reduced computational time compared to an FEM analysis [41]. For FEA analyses, simplified 3D geometries of the stator or even of the stator and the housing are often used for NVH assessment [87,90]. On the basis of vibration analyses, an acoustic analysis is often carried out in order to find corresponding sound pressure levels [90,91].

5.4.5. Rotordynamics

In ten papers, tasks belonging to rotordynamic analysis are carried out in the scope of the MDDA. The rotordynamic analyses are performed to ensure that the rotor is not operated in a critical speed range. Similar to the resonance effect in the context of NVH, the critical speed is the speed at which the rotor excites the structure at its natural frequency [32]. Strong vibrations that are capable of destroying the motor are the consequence of the resonance effect [92]. High-speed motors are operating close to their physical limits [93], which explains why almost all papers that consider rotordynamics in the context of MDDA address high-speed motors. As can be seen from Figure 10a, all three fidelity levels were identified for rotordynamic analyses. In [94], only the first-order natural frequency of the rotor is determined analytically and considered as part of the optimization objectives. A similar approach is applied in [37]. Reference [69] uses an analytical estimation of the critical speed based on a simplified rotor geometry. Reference [73] uses a lumped-parameter model by dividing the rotor into disk segments, allowing the incorporation of multiple diameters and material properties. Some authors make use of higher fidelity analyses for the rotordynamics analyses. Reference [95] uses a 3D-FEA of the rotor with a cooling fan impeller to determine the first bending mode of a high-speed PMSM. Reference [92] uses a similar approach. A few authors use a campbell diagram for the analysis and visualization of critical speed ranges [73,96].

It is noticeable that other tasks that belong to the mechanical domain in the development of electrical machines are only considered in single works in the scope of MDDAs in this review. These tasks include the design of non-active parts, such as the housing, end flanges or bearings. In [11,46], simple analytical calculations are used for this purpose in order to be able to provide a better estimation of the mass and dimensions of the motor, including the non-active parts.

5.4.6. Manufacturing

Another engineering domain analyzed in this review is manufacturing. Manufacturing influences on the design are often not calculated by separate calculations as it is performed for the other engineering domains. Papers considering manufacturing aspects use statistical models to account for manufacturing deviations. This circumstance prevents to analyze the manufacturing analysis fidelity, which is why no data regarding the fidelity of the manufacturing analysis were derived from the literature study. The manufacturing analyses encountered during this review are often referred to as robust design optimization. According to [97], there are three common approaches to conduct robust design for electric motors: Taguchi parameter design, worst case design and design for six sigma. Performing a multi-objective robust design study always involves a high computational cost, especially when high-fidelity analyses are used [97]. For this reason, most robust design papers in the study make use of methods to improve the computational efficiency of the analyses, like surrogate models [98,99]. Manufacturing aspects are considered almost exclusively in combination with electromagnetic analyses. Thus, multidomain robust design was not identified in this review. The only exception to this was found in [99]. The authors make use of a multi-physics model for the robust design of a SMC (soft magnetic composite) motor consisting of electromagnetic, thermal and modal analyses as well as a production cost estimation model. To handle the computational cost, surrogate models are built and a multilevel design parameter approach is employed.

5.4.7. Economics

The costs of a design also play a major role in many papers. This is reflected in the choice of optimization objectives in Figure 9a. In order to determine costs, different types of cost determination are used. According to [99], the production cost of electric motors consist of material cost, buying parts cost, machinery cost, capital cost and personnel cost. Papers in this review predominantly reduce this to the material cost and calculate it by computing the masses or volumes and the corresponding market prices [45,47,71,100,101].

In this review, almost all authors restrict costs, and thus mass and volume, to active material cost, i.e., iron sheets, copper parts and magnets [45,47,67]. Reference [100] also includes the cost of the rotor-retaining sleeve. References [101,102] include the aluminum parts of the motor, making a simplified housing geometry necessary. Reference [99] calculates the material cost in dependence of production volume and also include the cost of scrap material and buying parts in their estimation. References [77,103] are the only works that include machining cost in their calculation for manufacturing of a specific component: in this case, the SMC core.

5.4.8. Surrogate Models

Especially in the more recent papers, surrogate models are frequently implemented in MDDAs to handle expensive calculations with a lower time effort [84,100]. The use of surrogate models fundamentally affects the setup of the MDDA. The models must be built in advance of the actual optimization. The steps for this usually consist of sampling to build the database, building the surrogate model, and checking the accuracy of the model including subsequent readjustment [52,73,96,100]. For the sampling stage, different methods exist, e.g., the Latin hypercube sampling plan [90]. In [96], a design of experiments sampling plan is used to maximize the amount of information acquired and minimize the bias error. Reference [73] proposes to combine low- and high-fidelity domain analyses in the sampling stage to achieve the best trade-off between computation time and accuracy. The actual surrogate model can be built using either artificial neural networks or interpolation methods. In [100], three neural network methods, multilayer perceptron, support vector regression, generalized regression neural network, are compared with the Kriging interpolation method, where support vector regression shows the best results for the considered application of a high-speed PMSM.

5.4.9. Domain Couplings

Figure 12 presents the results of the domain couplings analysis. For this analysis, the rotordynamic, structural and NVH domains were considered as one mechanical domain. This is valid, since no couplings between different tasks within the mechanical domain were identified. The figure depicts the percentage distribution of coupling types in papers that consider the respective pairs of engineering domains. Couplings between domains are not always reported in the reviewed papers. In ten papers, no specific couplings or the type of coupling implemented in the respective MDDA are described. The coupling between electromagnetic and thermal domain is described as a strong coupling in the literature, making it necessary to be considered for accurate results [72]. In particular, the magnetic and electric conductivity of materials in the electric motor are depending on the temperature in the system [40]. One of the outputs of the electromagnetic calculations are losses, which are necessary inputs for the thermal domain in order to calculate the temperatures within the system [11,37,66]. Figure 12 shows that most authors that consider both the electromagnetic and thermal domain implement set coupling as a one-directional coupling. References [79,80] use a one-way coupling between the electromagnetic analysis and the thermal analysis to design cooling ducts in both the stator and the rotor. Reference [40] states that the permeability of the laminations remain nearly constant with changing temperature. The temperature dependency of copper resistivity on the other hand should be incorporated in the analysis. The authors propose to update the copper losses according to the temperature variation instead of recalculating the electromagnetic analysis [40]. If the initial guess of machine temperature deviates strongly from the calculated values, the calculation of the magnetic field and thus the magnetomotive force has to be updated [72,85]. This is why some authors implement a bi-directional coupling and iterate between the domains until the error becomes sufficiently small [76]. Reference [72] combines a one-way coupling with a two-way coupling using a space mapping methodology for the design of a starter motor.



Figure 12. Different types of couplings between engineering domains employed in the MDDA of reviewed papers.

The properties for the mechanical analyses also depend on the temperatures in the system. However, most papers that feature both a thermal and mechanical analysis do not consider a coupling between these domains. In [37], the temperature dependency is integrated into the mechanical analysis of magnets in a high-speed motor. The authors conclude that the temperature rise can have a significant impact on the safety factor of the magnets. Between electromagnetic and mechanical domains, different dependencies exist. The NVH analyses start with a vibrational analysis, which is based on radial electromotive forces, as an output of the electromagnetic analysis. This is why for papers featuring NVH analyses, a one-directional coupling between the electromagnetic analysis and NVH analyses is considered in all cases, e.g., [41,87,91]. For rotordynamics and structural analyses, dependencies arise mainly from rotational speeds in the system. Most MDDAs in this review base their calculations in these domains on the maximum rotational speed that the system is designed for [70,95]. In these cases, no direct coupling between the electromagnetic domain and rotordynamic/structural domain was identified. A number of papers designing detailed rotor geometries address coupling strategies between the electromagnetic and structural mechanics domains. It is necessary that the small ribs or bridges within its geometry, see Section 5.7, provide a balance between being sufficiently robust to withstand loads and yet small enough to reduce their detrimental electromagnetic influence [104]. Reference [104] presents and compares different architectures of coupled electromagnetic and mechanical analyses. Reference [83] suggests a parameter-split decoupling (PSD) approach, where the design parameters of a SynRM rotor are split into two subsets. One subset will be part of the electromagnetic analysis, with the other being fixed, and vice versa for the structural analysis. The assignment of a parameter to a subset depends on its influence in each domain. Most authors conduct the electromagnetic analyses and check for structural integrity afterwards in their MDDAs. However, the mechanical stress in the rotor laminations influences its permeability. In [105], an approach is presented where the results of a stress analysis in a PMSM rotor is considered in the electromagnetic analysis.

5.5. Applications

Figure 13a indicates that mobile applications dominate, such as electric vehicle (EV) traction motors, electric machines in aerospace applications or railway traction motors. However, not all articles provide details regarding the utilization of the motor developed in the MDDA. In most mobile applications, space is limited, which results in the aim to realize high power while keeping the weight or volume of the motor as low as possible [11]. The correlation between application and objectives becomes evident in Figure 13b. It indicates the objectives applied for papers focusing on traction motor design. Compared to the



numbers for all reviewed papers shown in Figure 9a, it can be concluded that losses, power density and dimensions of the motor are used more often as design objectives.



Reference [39] even states that for aerospace applications, minimizing the mass has the highest priority. Some papers for mobile applications are also focusing on optimizing specific driving cycles or load profiles. Reference [66] integrates a WLTP cycle (worldwide harmonized light vehicles test procedure) into the MDDA for an electric vehicle traction motor. With the calculated losses map, the total lost energy for the driving cycle is calculated and applied as the main optimization objective. Reference [39] uses a reference flight mission profile to calculate the total losses in the scope of their MDDA for a hybrid electric aircraft powertrain. A total of 21 of the reviewed papers feature high-speed motors. These papers do not, in all cases, specify an application field such as compressors or turbochargers. Consequently, they make up a large part of the "no application specified" group in Figure 13a.

High-speed motors are particularly noteworthy because of distinctive characteristics regarding the MDDA application. On the one hand, the application of MDDAs is quite apparent for these motors, since the high speeds are accompanied by high structural-mechanical, thermal and rotordynamic loads. Thus, sequential development processes for the development of high-speed motors comprise numerous iterations [5]. On the other hand, this type of motor entails changes in terms of the geometric design of the components. For example, because of high speeds, the motors are typically built with smaller diameters, and rotor-retaining sleeves must be provided to secure the magnets mechanically, see Section 5.7.

Furthermore, the application significantly impacts the considered domains of the MDDA. In Figure 14a,b, domain fidelity levels for high-speed motors and traction motors are given. For both applications, the designers opted to employ high-fidelity electromagnetic analyses. However, the structure of the MDDA varies significantly for all other domains. In the case of high-speed motors, the emphasis is on structural and rotodynamic analyses, with no consideration of NVH analyses. Conversely, traction motors are analyzed with respect to their thermal characteristics and NVH behavior.



Figure 14. (a) Considered domains and fidelity levels in papers designing high-speed motors; (b) Considered domains and fidelity levels in papers designing traction motors.

5.6. Process Stage

In Figure 15, it is illustrated that both preliminary and detailed design studies were identified in this review.



Figure 15. Share of papers presenting preliminary, detailed or two-staged design approaches.

Most of the papers focus on preliminary design aspects. In these papers, the multidisciplinary process primarily revolves around sizing the active machine part. This includes not only the main machine design parameters, but also the basic design of the stator and rotor geometries. More in-depth approaches start with fixed main machine design parameters and place emphasis on automating the design of more detailed geometries [66,83,104]. A small set of works combines a preliminary design step with a more detailed design [76,106]. From the bar graphs in Figure 16a,b, it can be seen that the process stage has a significant influence on the applied domain models and corresponding fidelity level. In preliminary design papers, all domains are present, drawing a similar picture as the findings in Figure 10a.



Figure 16. (a) Domains and corresponding fidelity levels of preliminary design papers; (b) Domains and corresponding fidelity levels of detailed design papers.

In more detailed designs, the focus of the analyses is on higher fidelity analysis models. For electromagnetic, structural and NVH analyses, FEA analyses are employed in the majority in MDDAs. For thermal analysis, LPMs are used for both preliminary and detailed design. Additionally, some detailed design papers tend to focus on component-level issues. Here, the analysis models are chosen depending on the component, e.g., structural and electromagnetic analyses for the design of PMSM rotor geometries [105] or SynRM rotors [43,65].

5.7. Type and Part of Electric Motor

5.7.1. Type of Electric Motor

As can be seen from Figure 17a, most authors (61.7%) apply MDDA to permanent magnet-excited motors. Overall, surface permanent magnet motors (SPM) predominate. The SPM motor type is often used for high-speed motors [8,55,73,75,100]. A general topology of SPM motors used in this review is shown in Figure 17b. A main feature of highspeed SPMs is that a rotor-retaining sleeve must be designed in the MDDA [37,60,70,75,95]. In [50], an MDDA approach is used to design an SPM for a traction motor application. In the process, optimal stator and rotor parameters are determined, including a rotorretaining sleeve. In general, MDDAs applied for the design of permanent magnet motors often include modules to examine permanent magnet-specific boundaries and properties: With permanent magnets, the examination of the temperature is of utmost importance to prevent demagnetization [42,70]. Furthermore, permanent magnets in the rotor structure can be exposed to high mechanical stresses, which is why structural mechanics modules might be necessary [37,95]. Papers dealing with interior permanent magnet motors (IPM) often consider rotor topologies with a V-magnet arrangement, as is depicted in Figure 17b exemplary [22,41,66]. For IPMs, the rotor topology is frequently studied in great detail, i.e., the majority of design variables in the MDDA describe rotor geometries [22,78,105]. Reference [66] determines the optimal shape and orientation of a V-type permanent magnet rotor. The embedding of permanent magnets often results in narrow rib and bridge geometries, which is why the authors implement a structural analysis of the rotor geometry. Similar approaches can be found in [22,78,105].

Next to PM motors, reluctance motors are the second largest group in this review, making up 20.2%. Synchronous reluctance motors account for 12.8%, and switched reluctance motors are considered in 7.4% of reviewed papers. Similar to the MDDA of IPMs, papers applying MDDAs for SynRM again focus on the design of the rotor geometry. SynRMs also comprise fine ribs or bridges, as can be seen in Figure 17b. The design of these has a significant electromagnetic and structural mechanical impact [44,68,104,106]. Almost

all induction motors (IMs) in the reviewed papers feature squirrel cages, an example of which is shown in Figure 17b. Consequently, the MDDAs in these papers often focus on the detailed design of the rotor bar geometry [101,107]. In addition to electromagnetic performance, the geometry of an IM rotor has an impact on thermal, structural, mechanical and NVH characteristics [81,93,108].



Figure 17. (**a**) Share of papers with applying MDDA for different types of electric motor topologies; (**b**) Typical geometries encountered in this review for the most frequently represented motor types.

Some authors intentionally keep their MDDA approach general, i.e., they do not refer specifically to one machine type. These papers thus claim that their approach can be applied to different types of electric motors [47,52,64,90]. One prerequisite for a general MDDA approach is, according to [47], to build the MDDA based on modular submodels, so that each can be replaced for the calculation of different machine types.

5.7.2. Automatically Designed Parts in the MDDA

Most reviewed papers are limited to the automated design of active parts, i.e., stator and rotor geometries. In some applications, the stator geometry is fixed and only an MDDA for the component specific parameters, e.g., rotor parameters, is performed. These papers take a more detailed look on the 2D-Geometry of PMSMs or SynRMs, often combining electromagnetic and structural mechanic calculations [66,83,104]. The majority of papers use simplified 2D geometries for analyses. If machine types are designed whose magnetic flux also has axial components, such as transverse flux machines or axial flux machines, 3D geometries are used [45,77,99]. Furthermore, the simplification of the motor geometry depends on the analyzed domains. In the case of NVH analyses, a simplified 3D geometry of either the stator and/or housing is often required, or a 2D simplification must be derived from the original geometry [41,88]. As mentioned previously, a significant number of papers focus on the design of high-speed motors. Consequently, it is common practice to incorporate a sleeve in addition to the typical stator and rotor geometries. This is because the presence of a sleeve directly affects the effective airgap in the magnetic circuit, thereby impacting essential optimization objectives such as power density and efficiency [55,75]. Non-active parts are rarely designed in the scope of an MDDA in the reviewed papers. In [94], magnetic bearings and the active parts are designed simultaneously as part of the MDDA for a high-speed motor. In [38,60,79,80], next to the active parts, a focus is put on designing geometries for the cooling system, e.g., cooling ducts in the stator or in the cooling

jacket. In some MDDA applications, the shaft is designed simultaneously with the active parts [37,51,60]. In [11,46,51], non-active parts are automatically designed, like bearings, shaft and housing parts, e.g., by generating cylindrical shells and disks. The objective here is to obtain a better estimation of the non-active mass in the motor and consider it within the optimization.

6. Findings

In the course of this review, the selected papers were examined with regard to twelve criteria. As stated in Section 4, the review criteria were established by considering application-unspecific design automation tasks and electric-motor-specific criteria. The papers showed significant differences with regard to most of the criteria examined, indicating that the criteria can be used to analyze MDDA processes for the development of electric motors. In the following sections, the key findings drawn from the review will be presented.

6.1. Mdda Application Is Limited to Specific Use Cases or Machines; General Works Are Lacking

Overall, this review has shown that MDDA is already being used in the development of electric motors. The results in Section 5.5 showed that MDDAs can be employed for the design of electric motors in a variety of applications. However, in 64% of reviewed papers, MDDAs are employed for mobile applications or high-speed motors. Thus, MDDAs are not yet adopted for electric motors in all application fields. It can be assumed that the reason for this is similar to that of setting up design automation for the design of other systems. Studies have shown that the initial effort to create the models and interfaces is considered to be the main obstacle and requires a corresponding change in the organization [109]. In this review, authors often construct process chains for their specific case without adhering to any specific methodology or utilizing preexisting concepts and architectures, like MDO architectures. Thus, general works that developers can apply as a blueprint for their use case to build an MDDA process chain, or even works that provide methods for building MDDA process chains, are lacking. However, Table 1 provides a summary of five papers that address parts of this issue.

Table 1. Selection of outstanding papers in this review regarding the application of MDDA for the development of electric motors.

Source	Considered Domains	Highlighted Features of the Approach
[39]	Thermal (Level 2), Electromagnetic (Level 1), Structural (Level 1)	Transient state thermal optimization, Optimization based on mission profiles, Component design (winding insulation)
[37]	Electromagnetic (Level 3), Thermal (Level 2), Structural (Level 1), Rotordynamics (Level 1)	High level of multidisciplinarity, Design variables based on sensitivity analysis Thermal/Structural coupling considered Response surface based optimization
[73]	Electromagnetic (Level 2 and 3), Structural (Level 1 and 3), Rotordynamics (Level 1 and 3)	Multi-fidelity surrogate model Component design (Rotor, shaft, sleeve)
[110]	Electromagnetic (Level 1), Thermal (Level 2), Structural (Level 1)	System and component aspects considered Component specific modeling Application independent approach Surrogate modeling
[11]	Electromagnetic (Level 1), Thermal (Level 2), Structural (Level 1)	Design of passive components considered Comparison of different motor topologies Application independent approach

Reference [39] introduces a comprehensive Multidisciplinary Design, Analysis and optimization (MDDA) approach for designing an electric motor as an integral component of a hybrid electric aircraft powertrain. The authors integrate higher-level design considerations to design the motor's active components. Simultaneously, they address lower-level design aspects encompassing windings insulation design by including partial discharge analysis, since this component showed to be most critical for their application. Simultaneously, they address component-level design aspects related to the insulation of windings and partial discharges, since the winding showed to be critical for their application. In terms of thermal analysis, the authors emphasize the criticality of accounting for application-specific load profiles within a transient thermal analysis framework, particularly for optimizing objectives dependent on mass.

In [37], an approach for the design of high-speed motors for antenna applications is presented. While application specific, the approach shows the integration of a variety of disciplines, considering electromagnetic, thermal, structural and rotordynamic aspects in the loop of the MDDA. The authors systematically compile their optimization objectives and design variables based on a sensitivity analysis. Additionally, they make use of a surrogate model based on a response surface methodology and validate their design by simulations and experiments.

Reference [73] focus solely on the design of the rotor in a high-speed PMSM in their MDDA approach, showing that MDDA approaches can be employed both on a higher level and a lower level. Herein, they combine electromagnetic, structural and rotordynamic analysis and employ a multi-fidelity surrogate model for the optimization.

While most works in this review divide the design problem in their MDDA according to the domains considered, Reference [110] decomposes the system based on its components, considering shaft, yoke, slots, winding and pole design issues. To find a system-level design, the analytical component models are integrated in to a system-level model by simplifying them using a surrogate model.

Reference [11] stands out because of their consideration of active and passive machine parts alike in their MDDA approach. They highlight the importance for the mass reduction of the machine. Additionally, their approach is the only one using the MDDA to compare different motor topologies, in this case internal, external, dual stator and dual rotor topologies.

Overall, from this review, it can be drawn that the architecture of an MDDA process chain is a complex task that demands the developer to consider a variety of different aspects. However, this review indicates a lack of overall guidance for developers on how to setup MDDA process chains.

Thus, the support to be developed on the one hand has to give a general framework that can be adopted for different applications. On the other hand, it has to address the gap of lacking methodological support for the setup of MDDAs for electric motors.

6.2. MDDAs Are Relevant Both on a System and Component Level

This review has pointed out that MDDAs can be employed both on the system level and for more detailed examinations of specific components on the component level. The latter is necessary to explore the boundaries of a design and investigate the effects of material or manufacturing innovations on a particular component, e.g., soft magnetic composite cores [77]. Furthermore, only a detailed analysis of a component allows to consider resulting design variants, manufacturing constraints or necessary component analyses, e.g., partial discharge analyses in the winding [39].

A strict separation or decomposition of system and component design for the electric motor are found only in a few papers (e.g., [110]). Frequently, all aspects are examined at a single level, which limits the potential for parallelization [111]. Therefore, a methodological framework for the MDDA setup should enable the designer to consider both system and component design aspects within the overall design and provide assistance for their integration and parallelization.

Furthermore, this review showed that MDDA is relevant both for the system-level design and for the detailed design of parts in electric motors. Subsequently, a general frame-

work has to consider both these aspects. For this type of framework, the methodological assistance should aid the developer in adopting concepts of integration or parallelization. Figure 18 presents the parts of the overall support concept that are based on the findings of this review.





6.3. Systematic Approaches for the Architecture of MDDAs of Electric Motors Are Lacking

This review has revealed a multitude of interdependencies for the MDDA setup: for the architecture of an MDDA, a variety of decisions have to be made, i.e., the consideration and fidelity of domain analyses, the way of reasoning to implement, the arrangement and coupling of the automated procedure, and resulting interfaces, as well as approaches for a higher computational efficiency. This review demonstrated that architecture decisions depend on the application, the resulting requirements and design objectives, as well as the specific type of electric motor and process stage employed. However, approaches to set up an MDDA workflow taking these into account are lacking.

6.3.1. Objectives

The selection of optimization objectives, design variables and constraints is a crucial task while setting up an MDDA workflow. It lies in the nature of multidisciplinary design that a vast amount of possible design variables, optimization objectives and constraints can be considered. Selecting too many objectives or design variables can result in a large computational burden [63]. Even though laying this foundation has a major influence on the outcomes of a design study, only a small number of authors use systematic approaches in order to decide for design variables and objectives (e.g., [37,57,99,100]).

6.3.2. Procedure Arrangement

The MDDAs in the various reviewed papers can be classified as MDOs. However, most authors do not specifically state that they set up an MDO in their work. This indicates that existing methods in the literature have not fully advanced to the field of designing electric motors. This is also reflected by the fact that the vast majority of papers set up their MDO as the most straight forward architecture, called multidisciplinary feasible (MDF). In this monolithic architecture, analyses are simply conducted after each other in each optimization loop. Other monolithic MDO architectures, like All-at-once (AAO) or individual discipline feasible (IDF), or even distributed architectures, like concurrent subspace optimization (CSSO), were not identified in the scope of this review. The only exceptions are the following references: [23,112]. Reference [112] compares different monolithic architectures for its MDDA of a PMSM, proving that remarkable wall clock savings can be achieved compared to MDF. Reference [23] states that their framework is able to decompose the original optimization problem into smaller subproblems, making use of distributed MDO architectures like CSSO. A major advantage of a hierarchical breakdown is that it enables the developer to solve multiple optimizations in parallel [111]. In this review, however, systematic approaches on selecting an architecture for the MDDA are lacking. In other

engineering disciplines, methods for this task exist, e.g., [113], who propose a question technique to determine MDO architectures for aerospace engineering.

6.3.3. Computational Efficiency Approaches

Adapting the MDO architecture is not the only way to achieve time savings. In the reviewed papers, surrogate-model-based approaches became more and more prevalent in recent years. In these approaches, analyses are conducted to gather data regarding the inputs and corresponding output responses first. Based on these data, a model is fitted, which can accurately predict the output across the input range. These models can then be used in the actual optimization of the system. Since surrogate models can be evaluated as fast as analytical models, this enables the designer to conduct optimizations with high accuracy and less time effort [84,100]. For the overall architecture of the MDDA, this means that the corresponding steps for setting up the surrogate models must be incorporated. Furthermore, it should be noted that the electric motor always is part of a system-of-systems context, meaning it is itself a component within a larger system, such as a powertrain in a vehicle [39]. Therefore, an MDDA framework should provide a basis for integration into higher-level design and optimization procedures. Surrogate model approaches can act as an enabler for this [110].

6.4. High Level of Multidisciplinarity Continues to Be the Exception

The utilization of MDDA is primarily motivated by the ability to consider interactions among engineering domains during the systems design in order to enhance performance and reduce time and cost in the design cycle [114]. To achieve these benefits, it is crucial to systematically include the relevant engineering domains within the MDDA framework. The reviewed papers primarily focused on the electromagnetic domain, which is evidently the most important domain and thus should always be considered in the system-level design. It is noticeable that either fully analytical models or FEA models were used. Across all papers reviewed, thermal analysis has also been shown to be a vital component of electric motor MDDA. Especially for power dense motors, the temperature is the main limiting factor [39]. For thermal analysis, LPMs are used almost exclusively at the system level. Structuralmechanical analyses were also prominently addressed, especially for PMSM or SynRM rotor geometries. Both high-fidelity FEA calculations and analytical models are employed within this domain. If the aim of the MDDA is to design the rotor geometry in more detail, the component design model should incorporate structural analyses. NVH analyses and rotordynamic analyses are clearly underrepresented compared to the other engineering domains. Nevertheless, this review pointed out that low-fidelity approaches to incorporate these domains with a small additional computational load do exist. Overall, this review showed that developers ranked the domains differently in terms of the trade-off between accuracy and computational cost. The majority of papers utilize FEA for electromagnetic analyses, whereas analytical and semi-analytical models are predominantly employed for other domains. Furthermore, this review showed that cost and mass estimation can play a major role for a wide range of developments; however, the evaluation of these is often restricted to active parts and thus are limited. For the general MDDA framework to consider rotordynamics and NVH analyses, geometric models of passive parts showed to be necessary, which then can also be used to obtain better estimation of cost and mass of the motor. Another aspect investigated in this review is the manufacturing domain. Overall, manufacturing-related topics are relatively scarce. In principle, MDDA approaches have the potential to take manufacturing aspects into account early in the development process. Robust design approaches are an example for this. However, the majority of these approaches are confined to investigating the interaction between the electromagnetic and manufacturing domains. Robust design analysis is also associated with potentially extensive computational requirements, restricting its application to only a small set of design variables and making the use of surrogate models necessary [99].

6.5. Knowledge for MDDAs Is Not Systematically Formalized, Limiting Reusability

As already laid out in Section 2, Reference [14] describes DA as an automation of sequential steps, while KBE involves a more systematic way of capturing engineering knowledge to emulate human decision making. Referring to this definition, this review revealed that MDDAs in the reviewed papers lack systematic ways of knowledge formalization. MDDAs mostly contain hard-coded knowledge in the form of geometries, topologies or procedural rules. Systematic ways of building knowledge bases for the setup of MDDAs were not found in this review. This limits the transferability and reusability of the approaches. Knowledge captured in knowledge based systems can be reused with CAD tools and simulation tools, reducing product development time and cost [115]. This is crucial, since one of the main factors stopping developers from setting up DAs is the initial effort required [109]. Systematic approaches to build up the knowledge formalization and knowledge representation [17], do not seem to not have found their way into the set up of MDDAs for electric motors. However, knowledge-based systems to support more conventional design approaches of electric motors exist.

Reference [116] presents a concept for building a knowledge-based system for the design of electric motors along the Verein Deutscher Ingenieure (VDI) VDI-2206 and VDI-2221 product development cycles.

Reference [117] presents a KBS to support the variant management of electric motors. By modeling variant and design knowledge, the system supports the designer by proposing design solutions in a semi-automatic manner.

6.6. Model Fidelity Decisions Depend on the Machine Type, the Application and the Process Stage

Analysis model decisions include the choice of domain models, their fidelity and couplings between them. This review, specifically Section 5.4, has shown that the application that the motor is designed for has a major influence on the MDDAs objectives and thus the analysis model decision. MDDA processes showed to be relevant at different stages in the spatial synthesis of a motor design. Therefore, MDDA processes should be adaptable to different process stages. Comparable multi-fidelity approaches have been presented in [64,73]. Consequently, in a systematic MDDA setup approach, the analysis model decision should be based on an analysis of the application and the process stage. This should be part of the systematic knowledge acquisition for the MDDA setup, which current approaches in this review are lacking. This review also revealed that analysis model decisions have an influence on the geometries to be generated. For example, while for the electromagnetic analysis, a 2D representation may be sufficient, for NVH analyses, simplified 3D representations might be necessary, as was discussed in Section 5.7.2.

6.7. Passive Parts Are Neglected in Most MDDAs

The review pointed out that a focus is put on the active parts of the electric motors in most MDDAs. However, this review also identified studies that emphasize the substantial impact of passive parts on specific optimization objectives, as discussed in Section 5.7.2. In addition to that, the synthesis of passive parts acts as an enabler for multidomain analyses, i.e., NVH analyses, rotordynamic and thermal analyses. This includes the shaft and bearings for rotordynamic analysis [37], as well as the housing for thermal [11] and NVH [88] analyses. Therefore, a general framework should be capable of considering these parts in addition to the active geometries. Moreover, this review showed that simple analytical models for multidomain calculations of these parts are already available. These suggest that the inclusion of non-active parts does not impose excessive computational burdens on the MDDA [37,41,51]. Structural analyses showed in this review to be motor-type specific, e.g., structural calculations for PMSM rotor geometries [22,36]. Furthermore, this review highlighted that the majority of MDDAs employ 2D geometric representations for electric motors. However, as already discussed for certain machine types, parts, and domain analyses, 3D geometries are necessary. Moreover, simplifying geometries for

analysis purposes can help to reduce computational effort [88]. The MDDA framework should therefore be able to generate the parts with a high fidelity and derive low-fidelity geometries for analysis purposes.

7. Methodological Framework

This review points out a lack of guidance to set up MDDA processes within the development of electric motors in a purposeful manner. The works analyzed partially show general and systematic approaches and frameworks, but approaches that are only valid for individual use cases. A more generic framework addressing different fields of applications and their requirements, the different levels (system level and component level) as well as the different design stages and tasks of design automation is missing. To support engineers in setting up MDDA process chains, we propose a methodological framework that integrates the findings of this review—see Section 6—and addresses all tasks and decisions needed for MDDA in electric motor development.

The forthcoming section presents the methodological framework's structure, as it is visualized in Figure 19. The framework comprises two main components: The system-level design model (SDM) and a component design model (CDM). In addition, the framework provides methodological support to prepare and adapt the models needed for the MDDA.



Figure 19. Methodological framework derived from the findings of this review.

7.1. System Design Model (SDM)

The system model describes a modular MDDA framework for the design of electric motors on system level. The aim is to give a general model of the motor that can be adapted to the specific circumstances, mainly depending on the application and the individual requirements, optimization objectives and design variables derived from the intended application of the electric motor. Moreover, the design process stage is specified with regard to the SDM. The SDM should be implemented in a modular way, so that it can be reused for new application scenarios. It represents a reference model of the electric motor and can be adapted to specific application scenarios, motor types and design process stages. This adaptation is enabled by exchanging or rearranging certain elements of the SDM, e.g., analysis models. This allows to reuse elements of the SDM, domain models that work for most applications have to be set up. These can be drawn from the results of this review, as they are presented in Section 5.4. The SDM features a central geometry generator, as

proposed in different MDO approaches from aerospace engineering such as [118]. This allows to generate geometries and geometry parameters for the different analyses, enabling a high number of domains to be considered in the MDDA [119]. In the SDM, this also includes passive geometries of the motor, which current approaches often neglect.

7.1.1. SDM Preperation and Setup

The setup preparation is performed in a preparation step. In this step, knowledge for the architecture of the MDDA and the creation of corresponding models are gathered. This includes: system-level requirements based on the application at hand, information regarding the process stage and the considered motor type. The support for this task can be synthesized from a variety of existing methods, e.g., generic requirements engineering methods [120] or electric-motor-specific requirements engineering methods [121]. Furthermore, the SDM preparation addresses the formalization not directly related to the geometry by using, e.g., ontologies. For electric motor design, approaches for knowledge formalization exist (e.g., [122]); however, they have not been applied in the context of the setup of MDDA processes.

Based on these preparations, the SDM is set up. This includes the architecture of the MDDA, domain model and coupling decisions, necessary geometric models, as well as the derivation of design variables, objectives and constraints. This review revealed that methods for this task have not found their way in to the setup of MDDA processes for electric motors. However, in other disciplines, methods exist for this task, especially from aircraft engineering (e.g., [113,123]). Reference [113] gathers knowledge regarding the MDO architecture before formalizing it in a semantic web. This is used to derive the MDDA architects for aircraft design to quickly arrange MDO workflows and visualize them using a extended design structure matrix (XDSM). These methods are promising to be adapted for the setup of architectures for the MDDA in electric motor development in future works.

7.1.2. Model Optimization

After the MDDA process is set up, sensitivity analyses can be employed to find the design variables with the highest impact on the optimization objectives. Suitable methods for this task were identified in this review, e.g., in [124], and can be integrated into the proposed framework. Furthermore, the SDM can be used to build surrogate models. This is useful to increase computational efficiency before performing the actual optimization or for integration into higher-level system models. Methods for this were also identified in this review, which can be reused and adopted for this framework, e.g., [96,100].

7.2. Component Design Model (CDM)

The component design model (CDM) is the second part of the support framework. The purpose of the CDM is to detail certain components, while integrating the results into the SDM. This allows a component of interest to be deeply analyzed and designed early in the process. This way, component-level aspects, like manufacturing constraints, can be systematically considered, which existing approaches lack. As can be seen from this review, the decision for detailed modeling of a component can be driven by various reasons. It ranges from examining the impact of a particular manufacturing method or material innovation for a component, e.g., SMC rotor cores [97], to detailed examination of a component for exploiting it for a specific use case, e.g., the detailed examination of rotor geometries [22,36].

7.2.1. CDM Preperation and Setup

Similar to the preparation step of the SDM, the preparation of the CDM setup includes a systematic acquisition and formalization of knowledge about system and process context of the component, manufacturing constraints, necessary evaluations of the component as well as possible variations of the component, e.g., different rotor topologies. Since the goal of the CDM is to automatically design a component, existing approaches from the literature can be adopted for this. Reference [10] proposes methods for the identification, evaluation and formalization of design automation tasks. The proposed templates could be adapted to structure the process of the CDM. Reference [125] presents a part decomposition approach for the design automation of flow components, systematically considering manufacturing constraints in the process. A similar approach could be applied for the systematic knowledge acquisition and formalization for the CDM preparation.

7.2.2. Architecture Adaptation

For the integration of the CDM, the architecture of the SDM needs to be adapted. Depending on the component, the CDM can be either integrated or parallelized. Reference [110] proposed in his multilevel approach an integration procedure by building surrogate models. Another integration approach can be a parameter split decoupling, as is demonstrated in [83]. This review showed that existing MDO architectures are not employed for electric motor design. In other engineering disciplines, different works that tackle the systematic derivation of MDO architectures which could be adapted for the framework exist, e.g., [113,123,126].

8. Conclusions

In this paper, a systematic literature review was conducted to examine the current state of the art and research in the multidisciplinary design automation (MDDA) of electric motors. A total of 94 papers were selected and analyzed, considering general criteria for the characterization of design automation tasks as well as criteria that are specific for the development and optimization of electric motors.

8.1. Answers to Research Questions

This review revealed that existing implementations of MDDA are often focusing on specific applications or design criteria of electric motors, while more universally approaches are missing. The answers to the research questions given below summarize the findings of this review:

- How is the multidomain analysis and automated geometry synthesis structured? The majority of MDDAs encountered in this review can be classified as MDOs. However, it is evident that authors do not refer to existing MDO architectures, known from, e.g., aircraft engineering. This is reflected in the fact that most authors employ straightforward multidisciplinary-feasible architectures (MDF); other MDO architectures are often not considered. Additionally, numerous analyses are carried out on a sole level without differentiation between architecture levels, such as the system and component levels. Overall, this review indicates a lack of systematic and reusable approaches that can be utilized to establish an MDDA based on the given use case.
- Which parts and geometries of electric motors are automatically created? A focus in the reviewed papers is put on active parts of electric motors. This review showed that MDDAs are applied at both the system and component levels. In the former, the active parts are usually optimized as a whole, while in the latter, a component is considered in detail, such as the optimization of a rotor geometry. This review also revealed a lack of consideration of passive parts of the motor, which are often essential for non-electromagnetic domains, e.g., NVH, structural, thermal or rotordynamics, as well as higher accuracy mass and cost analyses.
- When in the development process of electric motors are MDDA approaches being utilized?

This review points out that MDDA approaches are used at both the preliminary and detailed stages of the design process of electric motors. It has been shown that the process stage has a significant impact on the choice of domain models in MDDAs. Approaches that cover multiple stages or can adjust to them were found to be rare in the reviewed papers.

- How are domain-specific aspects and constraints formalized? In this review, shapebased and procedural rules were the only types of knowledge formalization used. Other expected forms for this category of design automation task, e.g., graph based formalization, object-oriented formalization and UML based formalization, were not identified for the setup of MDDAs.
- For which application and type of electric motor is MDDA already employed? The application of MDDA is mostly limited to high-speed drives and mobile applications, such as electric vehicle traction motors. This review demonstrated that the application has a significant impact on MDDA's architecture by affecting the selection of domain models and the objectives of the MDDA.
- Which domains are considered and implemented in the MDDA and at which fidelity level? MDDA process frameworks enabling the developer to analyze all relevant engineering domains remain the exception in the reviewed papers. Electromagnetic analyses build the basis of all MDDA process chains in this review, mostly being employed with high-fidelity models. Thermal analysis is the second-most frequently considered domain. Independent of the application and process stage, the majority of MDDAs adopted lumped-parameter models for thermal analyses. Structural analyses are mostly employed for detailed rotor geometries under high load, e.g., PMSM or SynRM geometries. NVH and rotordynamic analyses are often neglected in the reviewed papers.
- Which couplings between engineering domains are considered? Simple, one-directional couplings are usually implemented between the electromagnetic and other relevant domains for reasons of computational effort. The implementation of bi-directional couplings is almost exclusively implemented for the coupling between the electromagnetic and thermal domains and, even here, was rarely observed. The coupling between the electromagnetic and mechanical analyses are, if existent, one directional. Thus, the influence of mechanical loads and deformations on the electromagnetic analysis is neglected in most works. Couplings between thermal and mechanical analyses are mostly not considered. Instead, this physical coupling is represented in a simplified way by means of worst-case scenarios.

These findings are considered by different elements of the proposed framework for multidisciplinary design automation of electric motors introduced in Section 7.

8.2. Core Contributions

The reported systematic literature review expands the state of the art in the automated design of electric motors with the following contributions:

- 1. We present a set of twelve criteria to analyze the current multidisciplinary design automation approaches for designing electric motors. In addition to general criteria that characterize aspects such as knowledge representation or reasoning method, the classification criteria consider the type of electric motors, along with the related domains and their coupling. In particular, the specific criteria for electric motors can be used to build up MDDA process chains.
- 2. We provide a comprehensive overview of existing approaches in the field of multidisciplinary design automation for electric motors covering the period from 2007 to 2022. Based on this analysis, we indicate the main trends in the research field and limitations of existing approaches: consequent analysis and integration of domains, applicability and transferability of MDDA approaches, systematic incorporation of established MDO architectures, as well as the consideration of passive components in the motor.
- 3. Based on our assessment of the current literature and identified limitations, we suggest an initial methodological framework for establishing MDDA process chains tailored to specific applications, with a focus on integrating system and component-level design, as well as thorough domain consideration and coupling.

Following the design research methodology (DRM), the next research steps will focus on elaborating the methodological framework in more detail (prescriptive study), as well as validating its applicability through the development of electric motors within the stated framework (descriptive study II).

Author Contributions: Conceptualization: N.U. and D.I.; methodology: N.U. and D.I.; investigation: N.U. and K.W.; formal analysis: N.U.; writing original draft preparation: N.U. and K.W.; writing review and editing: D.I., K.W. and N.U. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Data Availability Statement: Not applicable.

Conflicts of Interest: The authors declare no conflict of interest.

Abbreviations

The following abbreviations are used in this manuscript:

AAO	All-at-once architecture	
ACO	Ant colony algorithm	
CAD	Computer-aided design	
CDM	Component design model	
CDS	Computer-based design synthesis	
CSSO	Concurrent subspace optimization architecture	
DA	Design automation	
DRM	Design research methodology	
EEC	Electric equivalent circuit	
EV	Electric vehicle	
FEA	Finite element analysis	
GA	Genetic algorithm	
IDF	Individual discipline feasible architecture	
IPM	Interior permanent magnet motor	
KBE	Knowledge-based engineering	
MDO	Multidisciplinary optimization	
MDDA	Multidisciplinary design automation	
LPM	Lumped-parameter model	
MDF	Multidisciplinary feasible	
MEC	Magnetic equivalent circuit	
MOPSO	Multi-objective particle swarm optimization	
NSGA	Non-dominated sorting genetic algorithm	
NVH	Noise, vibration, harshness	
PMSM	Permanent magnet synchronous motor	
PSD	Parameter split decoupling	
PSO	Particle swarm optimization	
RSM	Response surface methodology	
SDM	System design model	
SMC	Soft magnetic composite	
SQP	Sequential quadratic programming	
SynRM	Synchronous reluctance motor	
UML	Unified modeling language	
VDI	Verein Deutscher Ingenieure e.V.	
WLTP	Worldwide harmonized light vehicles test procedure	
XDSM	Extended design structure matrix	

Appendix A. Employed Search String

("design automation " OR "automat* design" OR "generat* design" OR "generative engineering" OR "Knowledge based Engineering" OR "Knowledge based System" OR KBE OR "design synthesis" OR optimis* OR optimiz*) AND (multi-physics OR "multi physics" OR "multi disciplinary" OR multidisciplinary OR multi-disciplinary OR multidomain OR "multi domain" OR holistic OR multi-domain) AND ((electric OR "permanent magnet" OR synchronous OR asynchronous OR induction OR reluctance OR AC OR DC) AND (Motor OR Machine OR Generator)).

References

- 1. EMSA. Electro Motor Systems Annex EMSA Overview 2019–2024. Available online: https://www.iea-4e.org/emsa/ (accessed on 14 September 2023).
- 2. Bramerdorfer, G.; Tapia, J.A.; Pyrhonen, J.J.; Cavagnino, A. Modern Electrical Machine Design Optimization: Techniques, Trends, and Best Practices. *IEEE Trans. Ind. Electron.* 2018, 65, 7672–7684. [CrossRef]
- Rechkemmer, S.K.; Izumi, R.; Zhang, W.; Sawodny, O. Estimation of Permanent Magnet Synchronous Machine Performance for Pre-Design Using a Reluctance Network. *IFAC PapersOnLine* 2018, *51*, 75–80. [CrossRef]
- Tüchsen, J. Konzeption, Entwicklung und Einführung des Assistenzsystems D-DAS für die Produktentwicklung Elektrischer Motoren. Ph.D. Thesis, FAU University Press, Erlangen, Germany, 2021. [CrossRef]
- Uzhegov, N.; Kurvinen, E.; Nerg, J.; Pyrhonen, J.; Sopanen, J.T.; Shirinskii, S. Multidisciplinary Design Process of a 6-Slot 2-Pole High-Speed Permanent-Magnet Synchronous Machine. *IEEE Trans. Ind. Electron.* 2016, 63, 784–795. [CrossRef]
- Xu, Z.; La Rocca, A.; Pickering, S.J.; Eastwick, C.; Gerada, C.; Bozhko, S. Mechanical and thermal design of an aeroengine starter/generator. In Proceedings of the 2015 IEEE International Electric Machines & Drives Conference (IEMDC), Coeur d'Alene, ID, USA, 11–13 May 2015; pp. 1607–1613. [CrossRef]
- Gerada, D.; Mebarki, A.; Brown, N.L.; Gerada, C. Optimal split ratio for high speed induction machines. In Proceedings of the 2010 IEEE Energy Conversion Congress and Exposition, Atlanta, GA, USA, 12–16 September 2010; pp. 10–16. [CrossRef]
- 8. Qin, X.F.; Shen, J.X. Split ratio optimisation of high-speed permanent magnet synchronous motor with multi-physics constraints. *IET Electr. Power Appl.* **2020**, *14*, 2450–2461. [CrossRef]
- Kalt, S.; Erhard, J.; Lienkamp, M. Electric Machine Design Tool for Permanent Magnet Synchronous Machines and Induction Machines. *Machines* 2020, *8*, 15. [CrossRef]
- 10. Rigger, E. Task Definition for Design Automation. Ph.D. Thesis, ETH, Zurich, Switzerland, 2019. [CrossRef]
- Golovanov, D.; Papini, L.; Gerada, D.; Xu, Z.; Gerada, C. Multidomain Optimization of High-Power-Density PM Electrical Machines for System Architecture Selection. *IEEE Trans. Ind. Electron.* 2018, 65, 5302–5312. [CrossRef]
- 12. Cederfeldt, M.; Elgh, F. Design Automation Systems in SMEs: Current State, Potential, Need and Requirements. In Proceedings of the 15th International Conference on Engineering Design, Melbourne, Australia, 15–18 August 2005.
- 13. Heikkinen, T. Multidisciplinary Design Automation: Working with Product Model Extensions. Ph.D. Thesis, Jönköping University, School of Engineering, Jönköping, Sweden, 2018.
- Van der Velden, C. Application of Knowledge Based Engineering Principles to Intelligent Automation Systems. Ph.D. Thesis, School of Aerospace Mechanical and Manufacturing Engineering College of Science, Engineering and Technology RMIT University, Melbourne, Australia, 2008.
- 15. Rigger, E.; Shea, K.; Stankovic, T. Task categorisation for identification of design automation opportunities. *J. Eng. Des.* **2018**, 29, 131–159. [CrossRef]
- 16. Verhagen, W.J.; Bermell-Garcia, P.; van Dijk, R.E.; Curran, R. A critical review of Knowledge-Based Engineering: An identification of research challenges. *Adv. Eng. Inf.* **2012**, *26*, 5–15. [CrossRef]
- Colombo, G.; Pugliese, D.; Klein, P.; Lutzemnberger, J. A study for neutral format to exchange and reuse engineering knowledge in KBE applications. In Proceedings of the 2014 International Conference on Engineering, Technology and Innovation (ICE), Bergamo, Italy, 23–25 June 2014; pp. 1–10. [CrossRef]
- Gulanová, J.; Gulan, L.; Forrai, M.; Hirz, M. Generative engineering design methodology used for the development of surfacebased components. *Comput.-Aided Des. Appl.* 2017, 14, 642–649. [CrossRef]
- 19. Regenwetter, L.; Nobari, A.H.; Ahmed, F. Deep Generative Models in Engineering Design: A Review. J. Mech. Des. 2022, 144, 071704. [CrossRef]
- Chakrabarti, A.; Shea, K.; Stone, R.; Cagan, J.; Campbell, M.; Hernandez, N.V.; Wood, K.L. Computer-Based Design Synthesis Research: An Overview. J. Comput. Inf. Sci. Eng. 2011, 11, 21003. [CrossRef]
- Gray, J.S.; Hwang, J.T.; Martins, J.R.R.A.; Moore, K.T.; Naylor, B.A. OpenMDAO: An open-source framework for multidisciplinary design, analysis, and optimization. *Struct. Multidisc. Optim.* 2019, 59, 1075–1104. [CrossRef]
- Lombard, P.; Soualmi, A.; Huang, L.; Rodriguez, A.; Leconte, V. A New Methodology to Design Electric Motors for Automotive Applications Including Magnetic, Thermal, Structural and Vibration Constraints. In Proceedings of the 2020 23rd International Conference on Electrical Machines and Systems (ICEMS), Hamamatsu, Japan, 24–27 November 2020; pp. 515–520. [CrossRef]
- 23. Ding, W.T.; An, L.X.; Wang, C.M.; Huang, Y.P.; Long, T.; Jiang, M.L. Multidisciplinary Integrated Simulation and Design Optimization Framework for Electromechanical Servo System. *Appl. Mech. Mater.* **2014**, *704*, 263–269. [CrossRef]
- 24. van Gent, I.; Aigner, B.; Beijer, B.; Jepsen, J.; La Rocca, G. Knowledge architecture supporting the next generation of MDO in the AGILE paradigm. *Prog. Aerosp. Sci.* 2020, 119, 100642. [CrossRef]

- Darmani, M.A.; Vaschetto, S.; Cavagnino, A. Multiphysic Design and Modeling of Rotating Electrical Machines. In Proceedings of the IECON 2021—47th Annual Conference of the IEEE Industrial Electronics Society, Toronto, ON, Canada, 13–16 October 2021; pp. 1–6. [CrossRef]
- Sun, Y.; Ma, Q.; EL-Refaie, A. Investigation of Coupling Mechanism of Multidisciplinary Sub-models of Electric Machines for Multi-physics optimization. In Proceedings of the 2020 International Conference on Electrical Machines (ICEM), Gothenburg, Sweden, 23–26 August 2020; pp. 204–210. [CrossRef]
- Lei, G.; Zhu, J.; Guo, Y.; Liu, C.; Ma, B. A Review of Design Optimization Methods for Electrical Machines. *Energies* 2017, 10, 1962. [CrossRef]
- Stipetic, S.; Miebach, W.; Zarko, D. Optimization in design of electric machines: Methodology and workflow. In Proceedings of the 2015 Intl Conference on Optimization of Electrical & Electronic Equipment (OPTIM) & 2015 Intl Symposium on Advanced Electromechanical Motion Systems (ELECTROMOTION), Side, Turkey, 2–4 September 2015; pp. 441–448. [CrossRef]
- 29. Blessing, L.T.; Chakrabarti, A. *DRM*, *a Design Research Methodology*, 1st ed.; Springer: Guildford, UK, 2009.
- 30. Kitchenham, B.; Pearl Brereton, O.; Budgen, D.; Turner, M.; Bailey, J.; Linkman, S. Systematic literature reviews in software engineering—A systematic literature review. *Inf. Softw. Technol.* **2009**, *51*, 7–15. [CrossRef]
- Page, M.J.; McKenzie, J.E.; Bossuyt, P.M.; Boutron, I.; Hoffmann, T.C.; Mulrow, C.D.; Shamseer, L.; Tetzlaff, J.M.; Akl, E.A.; Brennan, S.E.; et al. The PRISMA 2020 statement: An updated guideline for reporting systematic reviews. *BMJ* 2021, 372, n71. [CrossRef]
- 32. Tong, W. Mechanical Design of Electric Motors, 1st ed.; Taylor & Francis: Boca Raton, FL, USA, 2014.
- 33. Pyrhonen, J. Design of Rotating Electrical Machines, 2nd ed.; Wiley: Chichester, UK, 2014. [CrossRef]
- 34. Bracikowski, N.; Rossi, M.; Hecquet, M.; Gillon, F. Multi-physics design rules using lumped models for a permanent magnet synchronous machine. *Int. J. Appl. Electromagnet. Mech.* **2013**, *43*, 13–23. [CrossRef]
- Riviere, N.; Stokmaier, M.; Goss, J. An Innovative Multi-Objective optimization Approach for the Multiphysics Design of Electrical Machines. In Proceedings of the 2020 IEEE Transportation Electrification Conference & Expo (ITEC), Chicago, IL, USA, 23–26 June 2020; pp. 691–696. [CrossRef]
- Lin, R.; Sudhoff, S.D.; Nascimento, V.C.d. A Multi-Physics Design Method for V-Shape Interior Permanent-Magnet Machines Based on Multi-Objective Optimization. *IEEE Trans. Energy Convers.* 2020, 35, 651–661. [CrossRef]
- Islam, M.K.; Tasnim, K.N.; Choi, S.; Kwak, S.; Arafat, A. Designing High-Power Ultra-High-Speed Motor Using a New Multiphysics Multi-Objective Optimization Method for Mechanical Antenna Applications. *IEEE Access* 2022, 10, 106305–106323. [CrossRef]
- Dilevrano, G.; Ragazzo, P.; Ferrari, S.; Pellegrino, G.; Burress, T. Magnetic, Thermal and Structural Scaling of Synchronous Machines. In Proceedings of the 2022 IEEE Energy Conversion Congress and Exposition (ECCE), Detroit, MI, USA, 9–13 October 2022; pp. 1–8. [CrossRef]
- Pettes-Duler, M.; Roboam, X.; Sareni, B.; Lefevre, Y.; Llibre, J.F.; Fénot, M. Multidisciplinary Design Optimization of the Actuation System of a Hybrid Electric Aircraft Powertrain. *Electronics* 2021, 10, 1297. [CrossRef]
- Li, S.; Zhang, S.; Jiang, C.; Mayor, J.R.; Habetler, T.G.; Harley, R.G. A fast control-integrated and multiphysics-based multiobjective design optimization of switched reluctance machines. In Proceedings of the 2017 IEEE Energy Conversion Congress and Exposition (ECCE), Cincinnati, OH, USA, 1–5 October 2017; pp. 730–737. [CrossRef]
- 41. Song, T.; Liu, H.; Bu, B.; Zhang, Z. Rapid Calculation and Optimization of Vibration and Noise of Permanent-Magnet Synchronous Motors for EVs Based on Equivalent Structural Network. *Machines* **2022**, *10*, 281. [CrossRef]
- 42. Lim, S.; Min, S.; Hong, J.P. Optimal Rotor Design of IPM Motor for Improving Torque Performance Considering Thermal Demagnetization of Magnet. *IEEE Trans. Magn.* 2015, *51*, 1–5. [CrossRef]
- 43. Kuci, E.; Jansen, M.; Coulaud, O. Level set topology optimization of synchronous reluctance machines using a body-fitted mesh representation. *Struct. Mutltidiscip. Opt.* **2021**, *64*, 3729–3745. [CrossRef]
- Ram Kumar, R.M.; Vakil, G.; La Rocca, A.; Gerada, D.; Gerada, C.; Fernandes, B.G. Improving the Saliency of a High Speed SynRel Rotor by using High Strength Martensitic Sleeve. In Proceedings of the 2021 International Conference on Sustainable Energy and Future Electric Transportation (SEFET), Hyderabad, India, 21–23 January 2021; pp. 1–6. [CrossRef]
- 45. Nishanth, F.N.; Bohach, G.; Nahin, M.M.; van de Ven, J.; Severson, E.L. Development of an Integrated Electro-Hydraulic Machine to Electrify Off-highway Vehicles. *IEEE Trans. Ind. Appl.* **2022**, *58*, 6163–6174. [CrossRef]
- Golovanov, D.; Galassini, A.; Flanagan, L.; Gerada, D.; Xu, Z.; Gerada, C. Dual-Rotor Permanent Magnet Motor for Electric Superbike. In Proceedings of the 2019 IEEE International Electric Machines & Drives Conference (IEMDC), San Diego, CA, USA, 12–15 May 2019; pp. 951–956. [CrossRef]
- Dai, L.; Gao, J.; Zhang, W.; Huang, S. A Genetic-Taguchi Global Design Optimization Strategy for Surface-Mounted PM Machine. In Proceedings of the 2019 22nd International Conference on Electrical Machines and Systems (ICEMS), Harbin, China, 11–14 August 2019; pp. 1–6. [CrossRef]
- Pfister, P.D.; Perriard, Y. Very-High-Speed Slotless Permanent-Magnet Motors: Analytical Modeling, Optimization, Design, and Torque Measurement Methods. *IEEE Trans. Ind. Electron.* 2010, 57, 296–303. [CrossRef]
- Bouzidi, I.; Masmoudi, A.; Bianchi, N. Electromagnetic/thermal design procedure of an aerospace electric propeller. In Proceedings of the 2015 Tenth International Conference on Ecological Vehicles and Renewable Energies (EVER), Monte Carlo, Monaco, 31 March–2 April 2015; pp. 1–9. [CrossRef]

- Stephen Du, Z.; Tangudu, J.; Sur, A.; Soni, S. Multiphysics Design and Optimization for a SPM Machine with MnBi Magnet and 6.5% Silicon Steel Materials for Traction Applications. In Proceedings of the 2019 IEEE International Electric Machines & Drives Conference (IEMDC), San Diego, CA, USA, 12–15 May 2019; pp. 1001–1008. [CrossRef]
- Gerada, D.; Golovanov, D.; Xu, Z.; Papini, L.; Degano, M.; Zhang, H.; Gerada, C. Holistic electrical machine optimization for system integration. In Proceedings of the 2017 IEEE 3rd International Future Energy Electronics Conference and ECCE Asia (IFEEC 2017—ECCE Asia), Kaohsiung, Taiwan, 3–7 June 2017; pp. 980–985. [CrossRef]
- 52. Ghorbanian, V.; Mohammadi, M.H.; Ibrahim, I.; Lowther, D.A.; Hendershot, J. An HPC-Based Data-Driven Process for Design Exploration and Optimization of Motor Drives. In Proceedings of the 2019 IEEE International Electric Machines & Drives Conference (IEMDC), San Diego, CA, USA, 12–15 May 2019; pp. 597–602. [CrossRef]
- Sarikhani, A.; Mohammed, O. A multi-physics multi-objective optimal design approach of PM synchronous machines. In Proceedings of the 2014 International Conference on Electrical Machines (ICEM), Berlin, Germany, 2–5 September 2014; pp. 968– 974. [CrossRef]
- Semidey, S.A.; Duan, Y.; Mayor, J.R.; Harley, R.G.; Habetler, T.G. Optimal Electromagnetic-Thermo-Mechanical Integrated Design Candidate Search and Selection for Surface-Mount Permanent-Magnet Machines Considering Load Profiles. *IEEE Trans. Ind. Appl.* 2011, 47, 2460–2468. [CrossRef]
- 55. Zhao, W.; Wang, X.; Gerada, C.; Zhang, H.; Liu, C.; Wang, Y. Multi-Physics and Multi-Objective Optimization of a High Speed PMSM for High Performance Applications. *IEEE Trans. Magn.* **2018**, *54*, 1–5. [CrossRef]
- 56. Zhang, J.; Wang, H.; Zhu, S.; Lu, T. Multi-Physics Multi-Objective Optimal Design of Bearingless Switched Reluctance Motor Based on Finite-Element Method. *Energies* 2019, 12, 2374. [CrossRef]
- 57. Yan, W.; Chen, H.; Liu, X.; Ma, X.; Lv, Z.; Wang, X.; Palka, R.; Chen, L.; Wang, K. Design and multi–objective optimisation of switched reluctance machine with iron loss. *IET Electr. Power. Appl.* **2019**, *13*, 435–444. [CrossRef]
- Guo, Y.; Zhu, J.; Liu, D.; Lu, H.; Wang, S. Application of Multi-level Multi-domain Modeling in the Design and Analysis of a PM Transverse Flux Motor with SMC Core. In Proceedings of the 2007 7th International Conference on Power Electronics and Drive Systems (PEDS), Bangkok, Thailand, 27–30 November 2007; pp. 27–31. [CrossRef]
- Lopez, C.; Michalski, T.; Espinosa, A.; Romeral, L. New SynRM design approach based on behaviour maps analysis. In Proceedings of the 2016 XXII International Conference on Electrical Machines (ICEM), Lausanne, Switzerland, 4–7 September 2016; pp. 1915–1921. [CrossRef]
- Al-Ani, M.; Al-Timimy, A.; Giangrande, P.; Degano, M.; Lindh, P.; Zhang, H.; Galea, M.; Gerada, C. Influence of Manufacturing and Drive Effects in High-Speed, High-Power-Density PM Machine for Flooded Pump Application. In Proceedings of the 2021 IEEE Workshop on Electrical Machines Design, Control and Diagnosis (WEMDCD), Modena, Italy, 8–9 April 2021; pp. 88–94. [CrossRef]
- Schillingmann, H.; Heister, C.; Henke, M. Scaling process of synchronous reluctance machines for automotive applications. In Proceedings of the 2017 IEEE Transportation Electrification Conference and Expo (ITEC), Chicago, IL, USA, 22–24 June 2017; pp. 15–20. [CrossRef]
- 62. Boisson, J.; Louf, F.; Ojeda, J.; Mininger, X.; Gabsi, M. Analytical approach for magnetic and acoustic modeling of flux-switching permanent-magnet motors: Application to geometrical optimization. In Proceedings of the 2014 17th International Conference on Electrical Machines and Systems (ICEMS), Hangzhou, China, 22–25 October 2014; pp. 3060–3066. [CrossRef]
- Amrhein, M.; O'Connell, T.C.; Wells, J.R. An integrated design process for optimized high-performance electrical machines. In Proceedings of the 2013 International Electric Machines & Drives Conference, Chicago, IL, USA, 12–15 May 2013; pp. 847–854. [CrossRef]
- Kammermann, J.; Bertram, C.; Flugel, S.; Hecker, Q.; Meyer, W.; Herzog, H.G. Software based interaction of multiple domains for the design of electrical machines. In Proceedings of the 2014 17th International Conference on Electrical Machines and Systems (ICEMS), Hangzhou, China, 22–25 October 2014; pp. 624–629. [CrossRef]
- 65. Di Nardo, M.; Galea, M.; Gerada, C.; Palmieri, M.; Cupertino, F.; Mebarki, S. Comparison of multi-physics optimization methods for high speed synchrnous reluctance machines. In Proceedings of the IECON 2015—41st Annual Conference of the IEEE Industrial Electronics Society, Yokohama, Japan, 9–12 November 2015; pp. 2771–2776. [CrossRef]
- Martinez, I.; Palomar, D.J.; Garate, M.; Pardo, D.; Omella, Á.J.; Garcia, J.; Peña, A. A multi-physics approach for minimising motor losses on the WLTP driving cycle. In Proceedings of the 11th International Conference on Power Electronics, Machines and Drives (PEMD 2022), Hybrid Conference, Newcastle, UK, 21–23 June 2022; pp. 34–40. [CrossRef]
- 67. Akiki, P.; Hassan, M.H.; Bensetti, M.; Dessante, P.; Vannier, J.C.; Prieto, D.; McClelland, M. Multiphysics Design of a V-Shape IPM Motor. *IEEE Trans. Energy Convers.* 2018, 33, 1141–1153. [CrossRef]
- Kuci, E.; Henrotte, F.; Geuzaine, C.; Dehez, B.; de Greef, C.; Versele, C.; Friebel, C. Design Optimization of Synchronous Reluctance Machines for Railway Traction Application Including Assembly Process Constraints. In Proceedings of the 2020 International Conference on Electrical Machines (ICEM), Gothenburg, Sweden, 23–26 August 2020; pp. 117–123. [CrossRef]
- Palmieri, M.; Gallicchio, G.; Cupertino, F. An Automated Design Procedure for the Power Density Optimization of IPM machines. In Proceedings of the 2020 IEEE International Conference on Power Electronics, Drives and Energy Systems (PEDES), Jaipur, India, 16–19 December 2020; pp. 1–6. [CrossRef]

- Farhan, A.; Johnson, M.; Hanson, K.; Severson, E.L. Design of an Ultra-High Speed Bearingless Motor for Significant Rated Power. In Proceedings of the 2020 IEEE Energy Conversion Congress and Exposition (ECCE), Detroit, MI, USA, 11–15 October 2020; pp. 246–253. [CrossRef]
- 71. Makni, Z.; Besbes, M.; Marchand, C. A coupled electromagnetic–thermal model for the design of electric machines. *COMPEL Int. J. Comput. Math. Electr. Electron. Eng.* **2007**, *26*, 201–213. [CrossRef]
- 72. Vivier, S.; Friedrich, G. Comparison Between Single-Model and Multimodel Optimization Methods for Multiphysical Design of Electrical Machines. *IEEE Trans. Ind. Appl.* 2018, 54, 1379–1389. [CrossRef]
- 73. Xie, B.; Zhang, Y.; Xu, Z.; Zhang, F. Rotor Multidisciplinary Optimization of High Speed PMSM based on Multi-fidelity Surrogate Model and Gradient Sequential Sampling. *IEEE Trans. Energy Convers.* **2022**, *38*, 859–868. [CrossRef]
- Liu, H.; Wrobel, R.; Ayat, S.; Zhang, C. Coupled Electromagnetic and Thermal Design-Optimisation of a Traction IPM Machine with High- Torque Overload Capability. In Proceedings of the 2018 XIII International Conference on Electrical Machines (ICEM), Alexandroupoli, Greece, 3–6 September 2018; pp. 2647–2653. [CrossRef]
- Wu, S.; Huang, X.; Tian, C.; Zhang, P. Multi-Physical Field Optimization Analysis of High-Speed Permanent Magnet Synchronous Motor Based on NSGA-II Algorithm. In Proceedings of the 2019 22nd International Conference on Electrical Machines and Systems (ICEMS), Harbin, China, 11–14 August 2019; pp. 1–6. [CrossRef]
- Lopez-Torres, C.; Garcia Espinosa, A.; Riba, J.R.; Romeral, L. Design and Optimization for Vehicle Driving Cycle of Rare-Earth-Free SynRM Based on Coupled Lumped Thermal and Magnetic Networks. *IEEE Trans. Veh. Technol.* 2018, 67, 196–205. [CrossRef]
- Lei, G.; Liu, C.; Guo, Y.; Zhu, J. Multidisciplinary Design Analysis and Optimization of a PM Transverse Flux Machine With Soft Magnetic Composite Core. *IEEE Trans. Magn.* 2015, 51, 1–4. [CrossRef]
- Clauer, M.; Binder, A. Automated Fast Semi-Analytical Calculation Approach for the Holistic Design of a PMSM in a Novel Two-Drive Transmission. In Proceedings of the 2022 International Conference on Electrical Machines (ICEM), Valencia, Spain, 5–8 September 2022; pp. 1281–1287. [CrossRef]
- Schmidt, E.; Buschbeck, J.; Vogelsberger, M. Multi-physics optimization of air-cooled asynchronous induction machines for railway traction drives. In Proceedings of the 2017 20th International Conference on Electrical Machines and Systems (ICEMS), Sydney, NSW, Australia, 11–14 August 2017; pp. 1–6. [CrossRef]
- Vogelsberger, M.A.; Buschbeck, J.; Orellano, A.; Bazant, M.; Schmidt, E. Outcome of multi-physics optimization for increase in power density of asynchronous railway traction drives. In Proceedings of the 2015 17th European Conference on Power Electronics and Applications (EPE'15 ECCE-Europe), Geneva, Switzerland, 8–10 September 2015; pp. 1–8. [CrossRef]
- Chen, G.; Di, C.; Bao, X.; Xu, W. Calculation and Optimization of Thermal Stress in Induction Motor with Double-Skewed Rotor. In Proceedings of the 2021 24th International Conference on Electrical Machines and Systems (ICEMS), Gyeongju, Republic of Korea, 31 October–3 November 2021; pp. 1445–1449. [CrossRef]
- 82. Palmieri, M.; Cascella, G.L.; Cupertino, F. Design methodologies for the output power maximisation of synchronous reluctance machines. *IET Electr. Power Appl.* 2019, *13*, 1131–1140. [CrossRef]
- Di Nardo, M.; Lo Calzo, G.; Galea, M.; Gerada, C. Structural design optimization of a high speed synchronous reluctance machine. In Proceedings of the 2016 XXII International Conference on Electrical Machines (ICEM), Lausanne, Switzerland, 4–7 September 2016; pp. 2073–2079. [CrossRef]
- Godbehere, J.; Popescu, M.; Michon, M. Optimization of an IPM Traction Motor considering the Electric Drive Unit System Requirements. In Proceedings of the 2021 IEEE Energy Conversion Congress and Exposition (ECCE), Vancouver, BC, Canada, 10–14 October 2021; pp. 3667–3674. [CrossRef]
- 85. Bracikowski, N.; Hecquet, M.; Brochet, P.; Shirinskii, S.V. Multiphysics Modeling of a Permanent Magnet Synchronous Machine by Using Lumped Models. *IEEE Trans. Ind. Electron.* **2012**, *59*, 2426–2437. [CrossRef]
- Gundogmus, O.; Das, S.; Yasa, Y.; Elamin, M.; Sozer, Y.; Kutz, J.; Tylenda, J.; Wright, R.L. Acoustic Noise Mitigation in High Pole Count Switched Reluctance Machines Utilizing Skewing Method on Stator and Rotor Poles. *IEEE Trans. Ind. Electron.* 2022, 69, 5581–5593. [CrossRef]
- Bassi, E.; Vercesi, F.; Benzi, F. CAE Optimization of Synchronous Reluctance Motor and Vibro-Acoustic Analysis for Lift Systems. In Proceedings of the 2021 International Conference on Electromechanical and Energy Systems (SIELMEN), Iasi, Romania, 6–8 October 2021; pp. 139–143. [CrossRef]
- Chauvicourt, F.; Ramakrishnan, K.; Faria, C.; Myrria, G.; Mastinu, G. Analytical multi-physics methodology for fast acoustic noise prediction of an external rotor SPMSM. In Proceedings of the 2017 IEEE International Electric Machines and Drives Conference (IEMDC), Miami, FL, USA, 21–24 May 2017; pp. 1–6. [CrossRef]
- 89. Yu, Q.; Wang, X.; Cheng, Y. Multiphysics optimization design flow with improved submodels for salient switched reluctance machines. *Int. J. Appl. Electromagnet. Mech.* 2017, 54, 501–514. [CrossRef]
- 90. Lowther, D.; Ghorbanian, V.; Mohammadi, M.H.; Ibrahim, I. Design tools for electromagnetic-driven multi-physics systems using high performance computing. *COMPEL Int. J. Comput. Math. Electr. Electron. Eng.* **2019**, *39*, 198–205. [CrossRef]
- 91. Kim, H.; Nerse, C.; Lee, J.; Wang, S. Multidisciplinary Analysis and Multiobjective Design Optimization of a Switched Reluctance Motor for Improving Sound Quality. *IEEE Access* 2019, 7, 66020–66027. [CrossRef]

- Wang, S.; Tian, A.; Yang, W.; Song, D.; Bai, M. Rotor Design of HTS homopolar inductor alternator based on multi-physics field. In Proceedings of the 2021 International Conference on Power System Technology (POWERCON), Haikou, China, 8–9 December 2021; pp. 2451–2458. [CrossRef]
- Kurvinen, E.; Choudhury, T.; Narsakka, J.; Martikainen, I.; Sopanen, J.; Jastrzebski, R.P. Design Space Method for Conceptual Design Exploration of High Speed Slitted Solid Induction Motor. In Proceedings of the 2021 IEEE International Electric Machines & Drives Conference (IEMDC), Hartford, CT, USA, 17–20 May 2021; pp. 1–8. [CrossRef]
- Wang, X.; Xu, Y. Multi-Physics Multi-Objective Optimal Design of 5-DOF Magnetic Bearing System with High-speed Permanent Magnet Synchronous Motor. In Proceedings of the 2021 24th International Conference on Electrical Machines and Systems (ICEMS), Gyeongju, Republic of Korea, 31 October–3 November 2021; pp. 457–461. [CrossRef]
- 95. Du, G.; Huang, N.; Zhao, Y.; Lei, G.; Zhu, J. Comprehensive Sensitivity Analysis and Multiphysics Optimization of the Rotor for a High Speed Permanent Magnet Machine. *IEEE Trans. Energy Convers.* **2021**, *36*, 358–367. [CrossRef]
- 96. Dai, R.; Zhang, Y.; Wang, T.; Zhang, F.; Gerada, C.; Zhang, Y. Multi–objective optimisation of the HSPMM rotor based on the multi–physics surrogate model. *IET Electr. Power Appl.* **2021**, *15*, 1616–1629. [CrossRef]
- 97. Lei, G.; Bramerdorfer, G.; Ma, B.; Guo, Y.; Zhu, J. Robust Design Optimization of Electrical Machines: Multi-Objective Approach. *IEEE Trans. Energy Convers.* 2021, 36, 390–401. [CrossRef]
- Rafiee, V.; Faiz, J. Robust Design of an Outer Rotor Permanent Magnet Motor Through Six-Sigma Methodology Using Response Surface Surrogate Model. *IEEE Trans. Magn.* 2019, 55, 1–10. [CrossRef]
- 99. Ma, B.; Lei, G.; Zhu, J.; Guo, Y.; Liu, C. Application-Oriented Robust Design Optimization Method for Batch Production of Permanent-Magnet Motors. *IEEE Trans. Ind. Electron.* **2018**, *65*, 1728–1739. [CrossRef]
- 100. Gu, J.; Hua, W.; Yu, W.; Zhang, Z.; Zhang, H. Surrogate Models-based Multi-Objective Optimization of High-Speed PM Synchronous Machine: Construction and Comparison. *IEEE Trans. Transp. Electrif.* **2022**, *9*, 678–688. [CrossRef]
- 101. Le Besnerais, J.; Fasquelle, A.; Lanfranchi, V.; Hecquet, M.; Brochet, P. Mixed-variable optimal design of induction motors including efficiency, noise and thermal criteria. *Optim. Eng.* **2011**, *12*, 55–72. [CrossRef]
- Le Besnerais, J.; Hecquet, M.; Lanfranchi, V.; Brochet, P. Multi-objective optimization of the induction machine with minimization of audible electromagnetic noise. *Eur. Phys. J.*—*Appl. Phys.* 2007, *39*, 2007069. [CrossRef]
- Lei, G.; Zhu, J.; Liu, C.; Ma, B. Robust design optimization of electrical machines and drive systems for high quality mass production. In Proceedings of the 2016 6th International Electric Drives Production Conference (EDPC), Nuremberg, Germany, 30 November–1 December 2016; pp. 217–223. [CrossRef]
- 104. de Greef, C.; Kluyskens, V.; Henrotte, F.; Versele, C.; Geuzaine, C.; Dehez, B. Time-Efficient Multi-Physics Optimization Approaches for the Design of Synchronous Reluctance Motors. In Proceedings of the 2021 IEEE Energy Conversion Congress and Exposition (ECCE), Vancouver, BC, Canada, 10–14 October 2021; pp. 4197–4204. [CrossRef]
- 105. Yamazaki, K.; Kato, Y. Optimization of High-Speed Motors Considering Centrifugal Force and Core Loss Using Combination of Stress and Electromagnetic Field Analyses. *IEEE Trans. Magn.* **2013**, *49*, 2181–2184. [CrossRef]
- 106. Luise, F.; Tessarolo, A.; Agnolet, F.; Pieri, S.; Scalabrin, M.; Di Chiara, M.; de Martin, M. Design Optimization and Testing of High-Performance Motors: Evaluating a Compromise Between Quality Design Development and Production Costs of a Halbach-Array PM Slotless Motor. *IEEE Ind. Appl. Mag.* 2016, 22, 19–32. [CrossRef]
- 107. Rivière, N.; Popescu, M. Multi-physics Optimisation of a High Speed Copper Rotor Induction Motor for a Traction Application Using a Metamodel Based Approach. In Proceedings of the 10th International Conference on Power Electronics, Machines and Drives (PEMD 2020), Online Conference, 15–17 December 2020; pp. 461–466. [CrossRef]
- 108. Le Besnerais, J.; Fasquelle, A.; Hecquet, M.; Pelle, J.; Lanfranchi, V.; Harmand, S.; Brochet, P.; Randria, A. Multiphysics Modeling: Electro-Vibro-Acoustics and Heat Transfer of PWM-Fed Induction Machines. *IEEE Trans. Ind. Electron.* 2010, 57, 1279–1287. [CrossRef]
- R\u00e4deler, S.; Rigger, E. A Survey on the Challenges Hindering the Application of Data Science, Digital Twins and Design Automation in Engineering Practice. *Proc. Des. Soc.* 2022, 2, 1699–1708. [CrossRef]
- 110. Dai, Z.; Wang, L.; Meng, L.; Yang, S.; Mao, L. Multi-Level Modeling Methodology for Optimal Design of Electric Machines Based on Multi-Disciplinary Design Optimization. *Energies* 2019, *12*, 4173. [CrossRef]
- Kreuawan, S.; Moussouni, F.; Gillon, F.; Brisset, S.; Brochet, P.; Nicod, L. Optimal design of a traction system using Target Cascading. Int. J. Appl. Electromagnet. Mech. 2009, 30, 163–178. [CrossRef]
- Kreuawan, S.; Gillon, F.; Brochet, P. Optimal design of permanent magnet motor using multidisciplinary design optimization. In Proceedings of the 2008 18th International Conference on Electrical Machines (ICEM), Vilamoura, Portugal, 6–9 September 2008; pp. 1–6. [CrossRef]
- 113. Hoogreef, M. Advise, Formalize and Integrate MDO Architectures. Ph.D. Thesis, Delft University of Technology, Delft, The Netherlands, 2017. [CrossRef]
- 114. Martins, J.R.R.A.; Lambe, A.B. Multidisciplinary Design Optimization: A Survey of Architectures. AIAA J. 2013, 51, 2049–2075. [CrossRef]
- 115. Wu, D.; Gary Wang, G. Knowledge-Assisted Optimization for Large-Scale Design Problems: A Review and Proposition. *J. Mech. Des.* **2020**, *142*, 010801. [CrossRef]

- 116. Rivera, C.A.; Poza, J.; Ugalde, G.; Almandoz, G. A Knowledge Based System architecture to manage and automate the electrical machine design process. In Proceedings of the 2017 IEEE International Workshop of Electronics, Control, Measurement, Signals and their Application to Mechatronics (ECMSM), Donostia, Spain, 24–26 May 2017. [CrossRef]
- Tüchsen, J.; Dorsch, J.; Obendorfer, M.; Schleich, B.; Wartzack, S. Unterstützung des Variantenmanagements von Elektromotoren durch wissensbasierte Systeme. In Proceedings of the 30th Symposium Design for X (DFX 2019), Jesteburg, Germany, 18–19 September 2019. [CrossRef]
- 118. La Rocca, G.; van Tooren, M. Enabling distributed multi-disciplinary design of complex products: A knowledge based engineering approach. J. Des. Res. 2007, 5, 333. [CrossRef]
- 119. Sanchez, F.; Liscouët-Hanke, S.; Tfaily, A. Improving aircraft conceptual design through parametric CAD modellers—A case study for thermal analysis of aircraft systems. *Comput. Ind.* **2021**, *130*, 103467. [CrossRef]
- 120. Rupp, C. Requirements-Engineering und -Management: Aus der Praxis von Klassisch bis Agil, 6. aktualisierte und erweiterte auflage ed.; Hanser eLibrary; Carl Hanser Verlag GmbH & Co. KG: München, Germany, 2014. [CrossRef]
- 121. Rivera, C.A.; Poza, J.; Ugalde, G.; Almandoz, G. A Requirement Engineering Framework for Electric Motors Development. *Appl. Sci.* 2018, *8*, 2391. [CrossRef]
- 122. Mayr, A.; Meyer, A.; Schäffer, E.; Masuch, M.; von Lindenfels, J.; Mössinger, G.; Franke, J. Towards a Knowledge-Based Design Methodology for Managing the Complexity in the Integrated Product and Process Development of Electric Motors. In Advances in Production Research; Springer: Cham, Switzerland, 2019; pp. 112–125. [CrossRef]
- Page Risueño, A.; Bussemaker, J.; Ciampa, P.D.; Nagel, B. MDAx: Agile Generation of Collaborative MDAO Workflows for Complex Systems. In Proceedings of the AIAA AVIATION 2020 FORUM, Virtual Event, 15–19 June 2020. [CrossRef]
- 124. Shi, Z.; Sun, X.; Cai, Y.; Yang, Z. Robust Design Optimization of a Five-Phase PM Hub Motor for Fault-Tolerant Operation Based on Taguchi Method. *IEEE Trans. Energy Convers.* 2020, *35*, 2036–2044. [CrossRef]
- 125. Biedermann, M. Automated Design of Additive Manufactured Flow Components. Ph.D. Thesis, ETH Zurich, Zurich, Switzerland, 2022. [CrossRef]
- Yao, X.; Zhou, J.; Li, Y.; Liu, E. An Enhanced Collaborative Optimization Approach with Design Structure Matrix Algorithms to Group and Decouple Multidisciplines. *Math. Probl. Eng.* 2016, 2016, 4340916. [CrossRef]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.